\documentclass{ltxdoc}
\input{xepersian-doc-preamble.ltx}
\begin{document}
\section{Basics\label{basics}}
\subsection{Loading The Package}
You can load the package in the ordinary way;
\begin{BDef}
\Lcs{usepackage}\OptArgs\Largb{xepersian}
\end{BDef}
Where \texttt{options} of the package are explained later in \autoref{options}.

When loading the package, it is important to know that:
\begin{enumerate}
\item \textsf{xepersian} should be the last package that you load, because otherwise you are certainly going to overwrite \textsf{bidi} and \textsf{\XePersian} package's definitions and consequently, you will not get the expected output.
\item In fact, in addition to \textsf{bidi}, \textsf{\XePersian} also makes sure that some specific packages are loaded before \textsf{bidi} and \textsf{\XePersian}; these are those packages that \textsf{bidi} and \textsf{\XePersian} modifies them for bidirectional and Persian/English typesetting. 

If you load \textsf{\XePersian} before any of these packages, then you will get an error saying that you should load \textsf{\XePersian} or \textsf{bidi} as your last package. When it says that you should load \textsf{bidi} package as your last package, it really means that you should load \textsf{\XePersian} as your last package  as \textsf{bidi} package is loaded automatically by \textsf{\XePersian} package.

For instance, consider the following minimal example:
\begin{lstlisting}
\documentclass{minimal}
\usepackage{xepersian}
\usepackage{enumerate}
\settextfont{XB Niloofar}
\begin{document}
*\parsitext{Ř§ŰŮ ŮŮŘˇ ŰÚŠ Ř˘Ř˛ŮŘ§ŰŘ´ Ř§ŘłŘŞ}*
\end{document}
\end{lstlisting}
Where \textsf{enumerate} is loaded after \textsf{\XePersian}. If you run \texttt{xelatex} on this document, you will get an error which looks like this:
\begin{lstlisting}[numbers=none,backgroundcolor=\color{blue!30},frame=none,framexleftmargin=1mm]
! Package xepersian Error: Oops! you have loaded package enumerate after xepersian package. Please load package enumerate before xepersian package, and then try to run xelatex on your document again.

See the xepersian package documentation for explanation.
Type  H <return>  for immediate help.
 ...                                              
                                                  
l.5 \begin{document}
                    
? 

\end{lstlisting}
\end{enumerate}
\subsection{\textsf{\XePersian}'s Symbol}
 As you may know lion symbolizes \TeX{} but lion does not symbolizes \textsf{\XePersian}. \textbf{Simorgh}\footnote{\textbf{Simorgh} is an Iranian benevolent, mythical flying creature which has been shown on the titlepage of this documentation. For more details see \url{http://en.wikipedia.org/wiki/Simurgh}} (shown on the first page of this documentation) symbolizes \textsf{\XePersian}.
\subsection{Commands for Version number, and Date of The Package}
\begin{BDef}
\Lcs{xepersianversion}\quad\Lcs{xepersiandate}
\end{BDef}
\begin{itemize}
\item \Lcs{xepersianversion} gives the current version of the package.
\item \Lcs{xepersiandate} gives the current date of the package.
\end{itemize}
\begin{lstlisting}
\documentclass{article}
\usepackage{xepersian}
\settextfont{XB Niloofar}
\begin{document}
\begin{latin}
This is typeset by \textsf{\XePersian} package,\xepersianversion,
\xepersiandate.
\end{latin}
\end{document}
\end{lstlisting}
\subsection{{Options of The Package\label{options}}}
There are few options:
\subsubsection{\texttt{defaultlatinfontfeatures} Option}
\begin{BDef}
\texttt{defaultlatinfontfeatures}$=$\Larga{value}
\end{BDef}
The \texttt{defaultlatinfontfeatures} option allows you to pass  font features to the main Latin font. \Larga{value} is a set of font features.
\subsubsection{\texttt{extrafootnotefeatures} Option}
This is just the \texttt{extrafootnotefeatures} Option of \textsf{bidi} package. If you enable this option, you can typeset footnotes in paragraph form or in multi-columns (from two-columns to ten-columns). For more details, please read the manual of \textsf{bidi} package.
\subsubsection{\texttt{Kashida} Option}
If you pass \texttt{Kashida} option to the package, you will use Kashida for stretching words for better output quality and getting rid of underfull or overfull \Lcs{hbox} messages.
 
Note that you can not use \texttt{Kashida} option when you are using Nastaliq-like font (well, you still can use \texttt{Kashida} option when you use any Nastaliq-like font, but I can not guarantee high quality output!).

\bigskip
The following two commands are provided when you activate the \texttt{Kashida} option:
\begin{BDef}
\Lcs{KashidaOn}\quad\Lcs{KashidaOff}
\end{BDef}
\begin{itemize}
\item \Lcs{KashidaOn} enables Kashida and is active by default when \texttt{Kashida} option is activated.
\item \Lcs{KashidaOff} disables Kashida.
\end{itemize}

\subsubsection{\texttt{quickindex} Option}
When you generally want to prepare index for your Persian documents, you need to first run \texttt{xelatex}, then \texttt{xindy}, and again \texttt{xelatex} on your document respectively, which is very time consuming. The \texttt{quickindex} option gives you the index with only and only one run of \texttt{xelatex}. To use this feature, you will need to run \texttt{xelatex --shell-escape} on your \TeX{} document; otherwise you get an error which indicates that shell scape (or write18) is not enabled.

 This option is now obsolete and  equivalent to \texttt{quickindex-variant2} option.
\subsubsection{\texttt{quickindex-variant1} Option}
 Same as \texttt{quickindex} Option but uses variant one (in which \parsitext{Ř˘} is grouped under \parsitext{Ř§}) for sorting Persian alphabets. 
\subsubsection{\texttt{quickindex-variant2} Option}
 Same as \texttt{quickindex} Option but uses variant two (in which \parsitext{Ř˘} is a separate letter) for sorting Persian alphabets. 
\subsubsection{\texttt{localise} Option}
This option allows you to use most frequently-used \LaTeX\ commands and environments in Persian, almost like what \TeX-e-Parsi offers. This is still work in progress and we wish to add lots more Persian equivalents of \LaTeX\ and \TeX\ commands and environments. The Persian equivalents of \LaTeX\ and \TeX\ commands are shown in \autoref{lcs},  The Persian equivalents of \XePersian\ commands are shown in \autoref{xcs},  Persian equivalents of \LaTeX\  environments are shown in \autoref{lenv} and Persian equivalents of \XePersian\  environments are shown in \autoref{xenv}

Please note that the Persian equivalents of \LaTeX\ and \TeX\ commands and environments are only available after loading \textsf{xepersian} package. This means that  you  have to write all commands or environments that come before \Lcs{usepackage}\Largb{xepersian}, in its original form, i.e. \Lcs{documentclass}.

Not only you can use Persian equivalents of \LaTeX\ and \TeX\ commands and environments, but still original \LaTeX\ and \TeX\ commands and environments work too.

The \TeX\,  \LaTeX\, and \XePersian\  commands and environments and their Persian equivalents listed in \autoref{lcs}, \autoref{xcs}, \autoref{lenv} and \autoref{xenv} is not the whole story; If any command and environment in \autoref{lcs}, \autoref{xcs}, \autoref{lenv} and \autoref{xenv} have a starred version, their starred version also work. For example in \autoref{lcs}, the Persian equivalent of \Lcs{chapter}  is \Pcs{ŮŘľŮ}. I know that \Lcs{chapter} has a starred version, so this means \Pcs{ŮŘľŮ*} is also the Persian equivalent of \Lcs{chapter*}. Is that clear?

\medskip
However there is more; you can localise any other commands/environments you want. You can use the following commands to localise your own commands/environments: 
\begin{BDef}
\Lcs{eqcommand}\Largb{\Larga{command-name in Persian}}\Largb{\Larga{original  \LaTeX{} command-name}}\\
\Lcs{eqenvironment}\Largb{\Larga{environment-name in Persian}}\Largb{\Larga{original \LaTeX{} environment-name}}
\end{BDef}
 
\begin{center}
\tablecaption{The Equivalent \LaTeX\ and \TeX\ Commands\label{lcs}}
\tablehead
   {\bfseries Command in \TeX\ or \LaTeX\ &\bfseries  Equivalent Persian Command\\ \hline}
\tabletail
   {\hline \multicolumn{2}{r}{\emph{Continued on next page}}\\}
\tablelasttail{\hline}
\begin{supertabular}{lr}
\bicsintabular{Ř´ŮŘšâŘŹŘŻŮŮ}{@arstrut}
\bicsintabular{ŮŮŮ}{above}
\bicsintabular{ŮŘ§ŘľŮŮâÚŠŮŘŞŘ§ŮâŘ¨Ř§ŮŘ§ŰâŮŮŘ§ŰŘ´}{abovedisplayshortskip}
\bicsintabular{ŮŘ§ŘľŮŮâŘ¨Ř§ŮŘ§ŰâŮŮŘ§ŰŘ´}{abovedisplayskip}
\bicsintabular{ŘšŮŮŘ§ŮâÚÚŠŰŘŻŮ}{abstractname}
\bicsintabular{Ř§ÚŠŘłŮŘŞ}{accent}
\bicsintabular{ŮŘšŘ§Ů}{active}
\bicsintabular{Ř¨ŰŮŘ˛Ř§ŘłŘˇŘąŮŮŘąŘłŘŞ}{addcontentsline}
\bicsintabular{Ř§ŘśŘ§ŮŮâŘ¨ŘąŘŹŘąŰŮŮ}{addpenalty}
\bicsintabular{ŮŘ´Ř§ŮŰ}{address}
\bicsintabular{Ř¨ŰŮŘ˛Ř§Ř¨ŮâŮŮŘąŘłŘŞ}{addtocontents}
\bicsintabular{Ř§ŘśŘ§ŮŮâŘ¨ŘąŘ´ŮŘ§ŘąŮŘŻŮ}{addtocounter}
\bicsintabular{Ř§ŘśŘ§ŮŮâŘ¨ŘąŘ¨ŘšŘŻ}{addtolength}
\bicsintabular{Ř¨ŰŮŘ˛Ř§ŮŘśŘ§ŰâŮ}{addvspace}
\bicsintabular{ŘŞŮŘ¸ŰŮâŘ¨ŘŻŮŮŘ§ŰŰ}{adjdemerits}
\bicsintabular{Ř¨ŰŮŘ˛Ř§Ř¨Řą}{advance}
\bicsintabular{Ř¨ŘšŘŻŘ§Ř˛Ř§ŮŘŞŘłŘ§Ř¨}{afterassignment}
\bicsintabular{Ř¨ŘšŘŻŘ§Ř˛ÚŻŘąŮŮ}{aftergroup}
\bicsintabular{Ř§ŮŮ}{aleph}
\bicsintabular{ŘŽŘľŰŘľŮâŮŘłŘŞŘšŘ§ŘąŮŮŮ}{aliasfontfeature}
\bicsintabular{Ř§ŮŘŞŘŽŘ§Ř¨âŘŽŘľŰŘľŮâŮŘłŘŞŘšŘ§ŘąŮŮŮ}{aliasfontfeatureoption}
\bicsintabular{Ř´ÚŠŘłŘŞŮŰ}{allowbreak}
\bicsintabular{ŘŞŘŽŘľŰ@}{alloc@}
\bicsintabular{ŘŞŘŽŘľŰŘľâŰŘ§ŮŘŞŮ}{allocationnumber}
\bicsintabular{Ř´ÚŠŘłŘŞâŮŮŘ§ŰŘ´âŮŘŹŘ§Ř˛}{allowdisplaybreaks}
\bicsintabular{Ř­ŘąŮŮâŘ¨Ř˛ŘąÚŻ}{Alph}
\bicsintabular{Ř­ŘąŮŮâÚŠŮÚÚŠ}{alph}
\bicsintabular{ŮŘ§ŮâŮŮÚŮŰŮ}{alsoname}
\bicsintabular{Ů}{and}
\bicsintabular{Ř˛Ř§ŮŰŮ}{angle}
\bicsintabular{ŘšŮŮŘ§ŮâŮžŰŮŘłŘŞ}{appendixname}
\bicsintabular{ŘŞŮŘąŰŘ¨}{approx}
\bicsintabular{ŘšŘąŘ¨Ű}{arabic}
\bicsintabular{Ř˘ŘąÚŻ}{arg}
\bicsintabular{ŘąŮÚŻâŘŽŘˇâŘŹŘŻŮŮ}{arrayrulecolor}
\bicsintabular{ŮŘ§ŘľŮŮâŘłŘŞŮŮŮŘ§ŰâŘ˘ŘąŘ§ŰŮ}{arraycolsep}
\bicsintabular{ŘśŘŽŘ§ŮŘŞâŘŽŘˇâŘŹŘŻŮŮ}{arrayrulewidth}
\bicsintabular{ÚŠŘ´ŰŘŻÚŻŰâŘ˘ŘąŘ§ŰŮ}{arraystretch}
\bicsintabular{ŘŻŘąŘ˘ŘşŘ§Ř˛ŮŮŘ´ŘŞŘ§Řą}{AtBeginDocument}
\bicsintabular{ŘŻŘąŮžŘ§ŰŘ§ŮâŮŮŘ´ŘŞŘ§Řą}{AtEndDocument}
\bicsintabular{ŘŻŘąŘ§ŮŘŞŮŘ§ŰâŘˇŘ¨ŮŮ}{AtEndOfClass}
\bicsintabular{ŘŻŘąŘ§ŮŘŞŮŘ§ŰâŘłŘ¨ÚŠ}{AtEndOfPackage}
\bicsintabular{ŮŮŰŘłŮŘŻŮ}{author}
\bicsintabular{ŮŘˇŮŘ¨âŮžŘ´ŘŞ}{backmatter}
\bicsintabular{Ř´ÚŠŘ§ŮâŮžŘ´ŘŞ}{backslash}
\bicsintabular{Ř¨ŘŻŮŮŘ§ŰŰ}{badness}
\bicsintabular{ŮŰŮŮ}{bar}
\bicsintabular{ŮŘ§ŘľŮŮâÚŠŘąŘłŰ}{baselineskip}
\bicsintabular{ÚŠŘ´Ř´âŮŘ§ŘľŮŮâÚŠŘąŘłŰ}{baselinestretch}
\bicsintabular{ŮžŘąŘŻŘ§Ř˛Ř´âŘŻŘłŘŞŮâŘ§Ű}{batchmode}
\bicsintabular{Ř´ŘąŮŘš}{begin}
\bicsintabular{Ř´ŘąŮŘšâÚŮž}{beginL}
\bicsintabular{Ř´ŘąŮŘšâŘąŘ§ŘłŘŞ}{beginR}
\bicsintabular{Ř´ŘąŮŘšâÚŻŘąŮŮ}{begingroup}
\bicsintabular{ŮŘ§ŘľŮŮâÚŠŮŘŞŘ§ŮâŮžŘ§ŰŰŮâŮŮŘ§ŰŘ´}{belowdisplayshortskip}
\bicsintabular{ŮŘ§ŘľŮŮâŮžŘ§ŰŰŮâŮŮŘ§ŰŘ´}{belowdisplayskip}
\bicsintabular{ŘłŰŘ§Ů}{bf}
\bicsintabular{ŮžŰŘ´âŮŘąŘśâŘłŰŘ§Ů}{bfdefault}
\bicsintabular{Ř´ŮŘ§ŰŮâŘłŰŘ§Ů}{bfseries}
\bicsintabular{Ř´ŘąÚŻŘąŮŮ}{bgroup}
\bicsintabular{ŮŘąŘŹŮŘš}{bibitem}
\bicsintabular{ÚŠŘŞŘ§Ř¨âŮŘ§ŮŮ}{bibliography}
\bicsintabular{ŘłŘ¨ÚŠâÚŠŘŞŘ§Ř¨âŮŘ§ŮŮ}{bibliographystyle}
\bicsintabular{ŘšŮŮŘ§ŮâÚŠŘŞŘ§Ř¨âŮŘ§ŮŮ}{bibname}
\bicsintabular{ŮžŘąŘ´âŘ¨ŮŮŘŻ}{bigskip}
\bicsintabular{ŮŮŘŻŘ§ŘąŮžŘąŘ´âŘ¨ŮŮŘŻ}{bigskipamount}
\bicsintabular{ŘŽŘˇâŮžŘ§ŰŰŮâŘ´ŮŘ§ŮŘą}{botfigrule}
\bicsintabular{ŘšŮŘ§ŮŘŞâŮžŘ§ŰŰŮ}{botmark}
\bicsintabular{ÚŠŘ§ŘŻŘąŘŞŘ§ŮžŘ§ŰŰŮ}{bottompageskip}
\bicsintabular{ŮŘłŘ¨ŘŞâŮžŘ§ŰŰŮ}{bottomfraction}
\bicsintabular{ÚŠŘ§ŘŻŘą}{box}
\bicsintabular{Ř­ŘŻŘ§ÚŠŘŤŘąŘšŮŮâÚŠŘ§ŘŻŘą}{boxmaxdepth}
\bicsintabular{Ř¨Ř´ÚŠŮ}{break}
\bicsintabular{ÚŻŮŮŮŮ}{bullet}
\bicsintabular{ŘŻŮŮžŮ@ŮžŮŘŹ}{@cclv}
\bicsintabular{ŘŻŮŮžŮ@Ř´Ř´}{@cclvi}
\bicsintabular{Ř´ŘąŘ­}{caption}
\bicsintabular{ÚŠŘŻŘąŘŻŮ}{catcode}
\bicsintabular{ŘąŮŮŮŘ´ŘŞ}{cc}
\bicsintabular{ŮŘ§ŮâŘąŮŮŮŘ´ŘŞ}{ccname}
\bicsintabular{ŮŮŘˇŮâŮŘłŘˇ}{cdot}
\bicsintabular{ŮŮŘ§ŘˇâŮŘłŘˇ}{cdots}
\bicsintabular{ŘŞŮŘ¸ŰŮâŘ§Ř˛ŮŘłŘˇ}{centering}
\bicsintabular{ŘŽŘˇâŮŘłŘˇ}{centerline}
\bicsintabular{ÚÚŠ@Ů}{ch@ck}
\bicsintabular{ŮŘľŮ}{chapter}
\bicsintabular{ŘšŮŮŘ§ŮâŮŘľŮ}{chaptername}
\bicsintabular{ŮŮŰŘłŮ}{char}
\bicsintabular{ŘŞŘšŘąŰŮâŮŮŰŘłŮ}{chardef}
\bicsintabular{Ř¨ŘąŘłŰâŮŘąŮŘ§Ů}{CheckCommand}
\bicsintabular{ŮŘąŘŹŘš}{cite}
\bicsintabular{ŘŽŘˇŘ§ŰâŘˇŘ¨ŮŮ}{ClassError}
\bicsintabular{Ř§ŘˇŮŘ§ŘšâŘˇŘ¨ŮŮ}{ClassInfo}
\bicsintabular{ŮŘ´ŘŻŘ§ŘąŘˇŘ¨ŮŮ}{ClassWarning}
\bicsintabular{ŮŘ´ŘŻŘ§ŘąŘˇŘ¨ŮŮâŘ¨ŰâŘłŘˇŘą}{ClassWarningNoLine}
\bicsintabular{ŮŘ´Ř§ŮÚŻŘąŮŘąÚŠŘ˛Ű}{cleaders}
\bicsintabular{ŘŻŮŘľŮŘ­ŮâŮžŘ§ÚŠ}{cleardoublepage}
\bicsintabular{ŘľŮŘ­ŮâŮžŘ§ÚŠ}{clearpage}
\bicsintabular{ŘŽŘˇâŮŘ§ŮžŘą}{cline}
\bicsintabular{Ř¨Ř¨ŮŘŻŮŘąŮŘŻŰ}{closein}
\bicsintabular{Ř¨Ř¨ŮŘŻŘŽŘąŮŘŹŰ}{closeout}
\bicsintabular{Ř¨ŘłŘŞŮ}{closing}
\bicsintabular{ŘŹŘąŰŮŮâŘłŘąŘ¨ŮŘŻ}{clubpenalty}
\bicsintabular{ŘŽŘ§ŘŹ}{clubsuit}
\bicsintabular{ŘšŮŘ§ŮŘŞâŮžŘ§ŰŰŮâŘłŘŞŮŮâŘ§ŮŮ}{colbotmark}
\bicsintabular{ŘšŮŘ§ŮŘŞâŘ§ŮŮâŘłŘŞŮŮâŘ§ŮŮ}{colfirstmark}
\bicsintabular{ŘąŮÚŻ}{color}
\bicsintabular{ÚŠŘ§ŘŻŘąŘąŮÚŻ}{colorbox}
\bicsintabular{ŘšŮŘ§ŮŘŞâŘ¨Ř§ŮŘ§ŰâŘłŘŞŮŮâŘ§ŮŮ}{coltopmark}
\bicsintabular{ŘąŮÚŻâŘłŘŞŮŮ}{columncolor}
\bicsintabular{Ř¨ŰŮâŘłŘŞŮŮ}{columnsep}
\bicsintabular{ŮžŮŮŘ§ŰâŘłŘŞŮŮ}{columnwidth}
\bicsintabular{ŘŽŘˇâŘ¨ŰŮâŘłŘŞŮŮ}{columnseprule}
\bicsintabular{ŘłŘˇŘąŮŮŘąŘłŘŞ}{contentsline}
\bicsintabular{ŘšŮŮŘ§ŮâŮŮŘąŘłŘŞâŮŘˇŘ§ŮŘ¨}{contentsname}
\bicsintabular{ÚŠŮžŰ}{copy}
\bicsintabular{Ř­ŮâŘŞŘ§ŮŰŮ}{copyright}
\bicsintabular{Ř´ŮŘ§Řą}{count}
\bicsintabular{Ř´ŮŘ§Řą@}{count@}
\bicsintabular{ŘŞŘšŘąŰŮâŘ´ŮŘ§Řą}{countdef}
\bicsintabular{ŘłŘŽ}{cr}
\bicsintabular{ŘłŘŽâŘłŘŽ}{crcr}
\bicsintabular{ŮŘ§ŮâŮŘąŮŘ§Ů}{csname}
\bicsintabular{ÚŻŘ˛ŰŮŮâŘŹŘ§ŘąŰ}{CurrentOption}
\bicsintabular{ÚŠŘ§ŘŻŘąŘ¨ŰŮŘ§Ř¨ŰŮ}{dashbox}
\bicsintabular{Ř¨ŰŮŘ§Ř¨ŰŮâŘš}{dashv}
\bicsintabular{@ŘŞŘ§ŘąŰŘŽ}{@date}
\bicsintabular{ŘŞŘ§ŘąŰŘŽ}{date}
\bicsintabular{ŘąŮŘ˛}{day}
\bicsintabular{ŘŽŘˇâŮžŘ§ŰŰŮâŘ´ŮŘ§ŮŘąŮžŮŮ}{dblbotfigrule}
\bicsintabular{ŮŘłŘ¨ŘŞâŮžŮŮâŮžŘ§ŰŰŮ}{dblbottomfraction}
\bicsintabular{ŘŽŘˇâŘ¨Ř§ŮŘ§ŰâŘ´ŮŘ§ŮŘąŮžŮŮ}{dblfigrule}
\bicsintabular{ŮŘłŘ¨ŘŞâŘľŮŘ­ŮâŘ´ŮŘ§ŮŘąŮžŮŮ}{dblfloatpagefraction}
\bicsintabular{ŮŘ§ŘľŮŮâŘ¨ŰŮâŘ´ŮŘ§ŮŘąŮžŮŮ}{dblfloatsep}
\bicsintabular{ÚŠŘŻŮÚŠŘ§ŮâŘşŰŘąŮŮŘ§ŮŘˇŮŘą}{dblfntlocatecode}
\bicsintabular{ŮŘ§ŘľŮŮâŮŘŞŮâŮŘ´ŮŘ§ŮŘąŮžŮŮ}{dbltextfloatsep}
\bicsintabular{ŮŘłŘ¨ŘŞâŮžŮŮâŘ¨Ř§ŮŘ§}{dbltopfraction}
\bicsintabular{Ř§ŘšŮŘ§ŮâŮŮŮâŘŤŘ§Ř¨ŘŞ}{DeclareFixedFont}
\bicsintabular{Ř§ŘšŮŘ§ŮâŮžŘłŮŮŘŻÚŻŘąŘ§ŮŰÚŠ}{DeclareGraphicsExtensions}
\bicsintabular{Ř§ŘšŮŘ§ŮâŘŻŘłŘŞŮŘąÚŻŘąŘ§ŮŰÚŠ}{DeclareGraphicsRule}
\bicsintabular{Ř§ŘšŮŘ§ŮâŮŘąŮŘ§ŮâŮŮŮâŮŘŻŰŮŰ}{DeclareOldFontCommand}
\bicsintabular{Ř§ŘšŮŘ§ŮâÚŻŘ˛ŰŮŮ}{DeclareOption}
\bicsintabular{Ř§ŘšŮŘ§ŮâŮŘąŮŘ§ŮâŮŮŰ}{DeclareRobustCommand}
\bicsintabular{Ř§ŘšŮŘ§ŮâŮŮŮâŘšŮŘ§ŘŚŮ}{DeclareSymbolFont}
\bicsintabular{ŘŻŮŘąŘ¨ŘłŘŞŮ}{deadcycles}
\bicsintabular{ŘŞŘą}{def}
\bicsintabular{ŘŞŘšŘąŰŮ@ÚŠŮŰŘŻ}{define@key}
\bicsintabular{ŘŞŘšŘąŰŮâŘąŮÚŻ}{definecolor}
\bicsintabular{ŘŻŘąŘŹŮ}{deg}
\bicsintabular{ÚŠŘŻŘŹŘŻŘ§ŘłŘ§Ř˛}{delcode}
\bicsintabular{ŘŹŘŻŘ§ŘłŘ§Ř˛}{delimiter}
\bicsintabular{ŘśŘąŰŘ¨âŘŹŘŻŘ§ŘłŘ§Ř˛}{delimiterfactor}
\bicsintabular{ÚŻŮŘŻŰ}{depth}
\bicsintabular{ŘŽŘ´ŘŞ}{diamondsuit}
\bicsintabular{Ř§Ř¨ŘšŘ§ŘŻ}{dim}
\bicsintabular{Ř¨ŘšŘŻ}{dimen}
\bicsintabular{Ř¨ŘšŘŻ@}{dimen@}
\bicsintabular{Ř¨ŘšŘŻ@ŰÚŠ}{dimen@i}
\bicsintabular{Ř¨ŘšŘŻ@ŘŻŮ}{dimen@ii}
\bicsintabular{ŘŞŘšŘąŰŮâŘ¨ŘšŘŻ}{dimendef}
\bicsintabular{ŘŞŰŘąŮâÚŻŘ°Ř§ŘąŰ}{discretionary}
\bicsintabular{Ř´ÚŠŘłŘŞâŮŮŘ§ŰŘ´}{displaybreak}
\bicsintabular{ŘŞŮŘąŮŘŞÚŻŰâŮŮŘ§ŰŘ´}{displayindent}
\bicsintabular{ŘłŘ¨ÚŠâŮŮŘ§ŰŘ´}{displaystyle}
\bicsintabular{ŘšŘąŘśâŮŮŘ§ŰŘ´}{displaywidth}
\bicsintabular{ŘŞŮŘłŰŮ}{divide}
\bicsintabular{ŘˇŘ¨ŮŮâŮŮŘ´ŘŞŘ§Řą}{documentclass}
\bicsintabular{ÚŠŮ}{do}
\bicsintabular{ŘŞŘšŮŰŘśâÚŠŘŻŮŘ§}{dospecials}
\bicsintabular{ŮŮŘˇŮ}{dot}
\bicsintabular{ŮŮŘˇŮâŮŘłŘ§ŮŰ}{doteq}
\bicsintabular{ŮžŘąŮŮŘˇŮâŘ§}{dotfill}
\bicsintabular{ŮŮŘ§Řˇ}{dots}
\bicsintabular{ÚŠŘ§ŘŻŘąŘŻŮŮŘ§}{doublebox}
\bicsintabular{ŘąŮÚŻâŮŘ§ŘľŮŮâŘŻŮŘŽŘˇâŘŹŘŻŮŮ}{doublerulesepcolor}
\bicsintabular{ŮŘ§ŘľŮŮâŘ¨ŰŮâŘŻŮŘŽŘˇ}{doublerulesep}
\bicsintabular{ŮŮŘ´âŮžŘ§ŰŰŮ}{downarrow}
\bicsintabular{ŘšŮŮ}{dp}
\bicsintabular{ŘŞŘŽŮŰŮ}{dump}
\bicsintabular{ŘŞŘąÚŻ}{edef}
\bicsintabular{ŮžŘ§ÚŻŘąŮŮ}{egroup}
\bicsintabular{Ř§ŮŘŞŮŘ§ŰâŮŘ§ŘľŮŮ}{eject}
\bicsintabular{ÚŻŘąŮŮ}{else}
\bicsintabular{ŘŞŘ§ÚŠŰŘŻ}{em}
\bicsintabular{ÚŠŘ´Ř´âŮŘ§ŘŹŘąŮ}{emergencystretch}
\bicsintabular{ŮŮÚŠŘŻ}{emph}
\bicsintabular{@ŮžŮÚ}{@empty}
\bicsintabular{ŮžŮÚ}{empty}
\bicsintabular{ŮŘŹŮŮŘšŮâŮžŮÚ}{emptyset}
\bicsintabular{ŮžŘ§ŰŘ§Ů}{end}
\bicsintabular{ŮžŘ§ŰŘ§ŮâÚŮž}{endL}
\bicsintabular{ŮžŘ§ŰŘ§ŮâŘąŘ§ŘłŘŞ}{endR}
\bicsintabular{ŮžŘ§ŰŘ§ŮâŮŘ§ŮâŮŘąŮŘ§Ů}{endcsname}
\bicsintabular{ŮžŘ§ŰŘ§ŮâŘ§ŮŮŰŮâŘłŘą}{endfirsthead}
\bicsintabular{ŮžŘ§ŰŘ§ŮâŮžŘ§}{endfoot}
\bicsintabular{ŘŞŮâŘ¨ŮŘŻ}{endgraf}
\bicsintabular{ŮžŘ§ŰŘ§ŮâÚŻŘąŮŮ}{endgroup}
\bicsintabular{ŮžŘ§ŰŘ§ŮâŘłŘą}{endhead}
\bicsintabular{ŮžŘ§ŰŘ§ŮâŮŘąŮŘŻŰ}{endinput}
\bicsintabular{ŮžŘ§ŰŘ§ŮâŘ˘ŘŽŘąŰŮâŮžŘ§}{endlastfoot}
\bicsintabular{ÚŻŘłŘŞŘąŘ´âŘ§ŰŮâŘľŮŘ­Ů}{enlargethispage}
\bicsintabular{ŘŞŮâŘłŘˇŘą}{endline}
\bicsintabular{ŮŮŰŘłŮâŘŞŮâŘłŘˇŘą}{endlinechar}
\bicsintabular{Ř§ŮâŘŻŮŘąŰ}{enspace}
\bicsintabular{Ř§ŮâŮŘ§ŘľŮŮ}{enskip}
\bicsintabular{ŮŘąŮŘ§ŮâŘŹŘ§ŮŘ´ŰŮ}{eqcommand}
\bicsintabular{ŮŘ­ŰŘˇâŘŹŘ§ŮŘ´ŰŮ}{eqenvironment}
\bicsintabular{Ř§ŘąŘŹŘ§ŘšâŮŘą}{eqref}
\bicsintabular{ÚŠŮÚŠâŘŽŘˇŘ§}{errhelp}
\bicsintabular{ŮžŰŘ§ŮâŘŽŘˇŘ§}{errmessage}
\bicsintabular{ŘłŘˇŘąŮŘŞŮâŘŽŘˇŘ§}{errorcontextlines}
\bicsintabular{ŮžŘąŘŻŘ§Ř˛Ř´âŘŞŮŮŮâŘŽŘˇŘ§}{errorstopmode}
\bicsintabular{ŮŮŰŘłŮâŮŰÚŮ}{escapechar}
\bicsintabular{ŰŮŘąŮ}{euro}
\bicsintabular{Ř­Ř§Ř´ŰŮâŘ˛ŮŘŹ}{evensidemargin}
\bicsintabular{ŮŘąŘłŘŽ}{everycr}
\bicsintabular{ŮŘąŮŮŘ§ŰŘ´}{everydisplay}
\bicsintabular{ŮŘąÚŠŘ§ŘŻŘąŘ§}{everyhbox}
\bicsintabular{ŮŘąÚŠŘ§Řą}{everyjob}
\bicsintabular{ŮŘąŘąŰŘ§ŘśŰ}{everymath}
\bicsintabular{ŮŘąŘ¨ŮŘŻ}{everypar}
\bicsintabular{ŮŘąÚŠŘ§ŘŻŘąŮ}{everyvbox}
\bicsintabular{Ř§ŘŹŘąŘ§ŰâÚŻŘ˛ŰŮŮâŮŘ§}{ExecuteOptions}
\bicsintabular{ŘŹŘąŰŮŮâŘ§ŘśŘ§ŮŰâŘŞŰŘąŮâŘ¨ŮŘŻŰ}{exhyphenpenalty}
\bicsintabular{Ř¨ÚŻŘłŘŞŘąŮžŘłâŘ§Ř˛}{expandafter}
\bicsintabular{ŮŘ§ŘľŮŮâŘ§ŘśŘ§ŮŰâŘ¨ŰŮâŘłŘŞŮŮŮŘ§}{extracolsep}
\bicsintabular{@Ř§ŮŮŰâŘ§Ř˛ŰÚŠ}{@firstofone}
\bicsintabular{@Ř§ŮŮŰâŘ§Ř˛ŘŻŮ}{@firstoftwo}
\bicsintabular{Ú@Ř§Řą}{f@ur}
\bicsintabular{ŘŽŘ§ŮŮŘ§ŘŻŮ}{fam}
\bicsintabular{ŘľŮŘ­ŮâŘŞŘŹŮŮŰ}{fancypage}
\bicsintabular{ÚŠŘ§ŘŻŘąŘ¨Ř§}{fbox}
\bicsintabular{ŘśŘŽŘ§ŮŘŞâÚŠŘ§ŘŻŘąŘ¨Ř§}{fboxrule}
\bicsintabular{Ř­Ř§Ř´ŰŮâÚŠŘ§ŘŻŘąŘ¨Ř§}{fboxsep}
\bicsintabular{ÚŠŘ§ŘŻŘąŘ¨Ř§ŘąŮÚŻ}{fcolorbox}
\bicsintabular{ŘąÚŻ}{fi}
\bicsintabular{ŘšŮŮŘ§ŮâŘ´ÚŠŮ}{figurename}
\bicsintabular{ŮžŘąŘ´ÚŠŮ}{filbreak}
\bicsintabular{ŮžŘą}{fill}
\bicsintabular{ŘšŮŘ§ŮŘŞâŘ§ŮŮ}{firstmark}
\bicsintabular{ŮžŮŮ}{flat}
\bicsintabular{ŮŘłŘ¨ŘŞâŘľŮŘ­ŮâŘ´ŮŘ§ŮŘą}{floatpagefraction}
\bicsintabular{ŘŹŘąŰŮŮâŘ´ŮŘ§ŮŘą}{floatingpenalty}
\bicsintabular{ŮŘ§ŘľŮŮâŘ¨ŰŮâŘ´ŮŘ§ŮŘą}{floatsep}
\bicsintabular{ŘŞŮŘ¸ŰŮâŘ§Ř˛ŮžŘ§ŰŰŮ}{flushbottom}
\bicsintabular{Ř´ÚŠŮŘ¨ŮŘŻŰ}{fmtname}
\bicsintabular{ŘąŘŻŮâŘ´ÚŠŮŘ¨ŮŘŻŰ}{fmtversion}
\bicsintabular{ŮŘ´Ř§ŮŮ}{fnsymbol}
\bicsintabular{ŮŮŮ}{font}
\bicsintabular{Ř¨ŘšŘŻŮŮŮ}{fontdimen}
\bicsintabular{ŘąŮŘ˛ŰŮŮâŮŮŮ}{fontencoding}
\bicsintabular{ŮŘ§ŮŰŮâŮŮŮ}{fontfamily}
\bicsintabular{ŮŘ§ŮâŮŮŮ}{fontname}
\bicsintabular{Ř´ŮŘ§ŰŮâŮŮŮ}{fontseries}
\bicsintabular{Ř´ÚŠŮâŮŮŮ}{fontshape}
\bicsintabular{Ř§ŮŘŻŘ§Ř˛ŮâŮŮŮ}{fontsize}
\bicsintabular{Ř¨ŮŮŘŻŘ§ŰâŮžŘ§ŰŰŮâŘľŮŘ­Ů}{footheight}
\bicsintabular{ŘŻŘąŘŹâŘ˛ŰŘąŮŮŰŘł}{footins}
\bicsintabular{Ř˛ŰŘąŮŮŰŘł}{footnote}
\bicsintabular{ŘšŮŘ§ŮŘŞâŘ˛ŰŘąŮŮŰŘł}{footnotemark}
\bicsintabular{ŘŽŘˇâŘ˛ŰŘąŮŮŰŘł}{footnoterule}
\bicsintabular{ŮŘ§ŘľŮŮâŘŞŘ§Ř˛ŰŘąŮŮŰŘł}{footnotesep}
\bicsintabular{Ř§ŮŘŻŘ§Ř˛ŮâŘ˛ŰŘąŮŮŰŘł}{footnotesize}
\bicsintabular{ŮŘŞŮâŘ˛ŰŘąŮŮŰŘł}{footnotetext}
\bicsintabular{ŮŘ§ŘľŮŮâŘŞŘ§ŮžŘ§ŰŰŮâŘľŮŘ­Ů}{footskip}
\bicsintabular{ŮŘąŰŮ}{frame}
\bicsintabular{ÚŠŘ§ŘŻŘąŘ¨Ř§ŘŽŘˇ}{framebox}
\bicsintabular{ŮŮŘ§ŘľŮâŰÚŠŮŮŘ§ŘŽŘŞâŮŘ§ŘŞŰŮ}{frenchspacing}
\bicsintabular{ŮŘˇŮŘ¨âŮžŰŘ´}{frontmatter}
\bicsintabular{Ř¨ŘšŘŻŘ¨ÚŻŘ°Ř§Řą}{futurelet}
\bicsintabular{@ŘŽŮŘąŘ­ŘąŰŘľŘ§ŮŮ}{@gobble}
\bicsintabular{@ŘŽŮŘąŘ­ŘąŰŘľŘ§ŮŮâŘŻŮ}{@gobbletwo}
\bicsintabular{@ŘŽŮŘąŘ­ŘąŰŘľŘ§ŮŮâÚŮŘ§Řą}{@gobblefour}
\bicsintabular{@ŘšŘ§ŮŘŞâŘ˘}{@gtempa}
\bicsintabular{@ŘšŘ§ŮŘŞâŘ¨}{@gtempb}
\bicsintabular{ŘŞŘąŘš}{gdef}
\bicsintabular{Ř§ŮÚŻŮŰâŘ§ŘˇŮŘ§Řš}{GenericInfo}
\bicsintabular{Ř§ŮÚŻŮŰâŮŘ´ŘŻŘ§Řą}{GenericWarning}
\bicsintabular{Ř§ŮÚŻŮŰâŘŽŘˇŘ§}{GenericError}
\bicsintabular{ŘšŘ§Ů}{global}
\bicsintabular{ŘŞŘšŘ§ŘąŰŮâŘšŘ§Ů}{globaldefs}
\bicsintabular{ŮŘşŘŞâŮŘ§ŮŮ}{glossary}
\bicsintabular{ŮŮŘąŮâŮŘąŮŮÚŻ}{glossaryentry}
\bicsintabular{ŘŽŮŘ´âŘ´ÚŠŮ}{goodbreak}
\bicsintabular{ÚŠŘ§ŘşŘ°ÚŻŘąŘ§Ů}{graphpaper}
\bicsintabular{ÚŻŰŮŮŮâÚŮž}{guillemotleft}
\bicsintabular{ÚŻŰŮŮŮâŘąŘ§ŘłŘŞ}{guillemotright}
\bicsintabular{ÚŻŰŮŮŮâŘŞÚŠŰâÚŮž}{guilsinglleft}
\bicsintabular{ÚŻŰŮŮŮâŘŞÚŠŰâŘąŘ§ŘłŘŞ}{guilsinglright}
\bicsintabular{ŘąŘŻŰŮâŘ§}{halign}
\bicsintabular{Ř¨ŘąŮŘŞŮ}{hang}
\bicsintabular{Ř¨ŘšŘŻŘ§Ř˛ŘłŘˇŘą}{hangafter}
\bicsintabular{ŘŞŮŘąŮŘŞÚŻŰâŘŤŘ§Ř¨ŘŞ}{hangindent}
\bicsintabular{Ř¨ŘŻŮŮŘ§ŰŰâŘ§}{hbadness}
\bicsintabular{ÚŠŘ§ŘŻŘąŘ§}{hbox}
\bicsintabular{Ř¨ŮŮŘŻŘ§ŰâŘłŘąŘľŮŘ­Ů}{headheight}
\bicsintabular{ŮŘ§ŘľŮŮâŘ§Ř˛ŘłŘąŘľŮŘ­Ů}{headsep}
\bicsintabular{ŘłŘąŘ¨ŮâŮŘ§Ů}{headtoname}
\bicsintabular{ŘŻŮ}{heartsuit}
\bicsintabular{Ř¨ŮŮŘŻŘ§}{height}
\bicsintabular{ŮžŘąŘ§}{hfil}
\bicsintabular{ŮžŘąŘąŘ§}{hfill}
\bicsintabular{ŘąŮŘšâŮžŘąŘ§}{hfilneg}
\bicsintabular{ŮžŘąŘ˛Ř§ŮŮŰ}{hfuzz}
\bicsintabular{ŮŘ§ŘľŮŮâŮŘŽŮŰ}{hideskip}
\bicsintabular{ŘšŘąŘśâŮžŮŮŘ§Ů}{hidewidth}
\bicsintabular{ŘŽŘˇâŮžŘą}{hline}
\bicsintabular{Ř­Ř§Ř´ŰŮâŘ§}{hoffset}
\bicsintabular{Ř­ŮŘ¸âŘŻŘąŘŹ}{holdinginserts}
\bicsintabular{ŮŘ§ŘľŮŮâŘ§ÚŻŘąŘŻ}{hrboxsep}
\bicsintabular{ŘŽŘˇâŘ§}{hrule}
\bicsintabular{ŮžŘąŘŽŘˇâŘ§}{hrulefill}
\bicsintabular{ŘˇŮŮâŘłŘˇŘą}{hsize}
\bicsintabular{ŮŘ§ŘľŮŮâŘ§}{hskip}
\bicsintabular{ŮŘśŘ§ŰâŘ§}{hspace}
\bicsintabular{ŮŘąŘŻŮŘ§}{hss}
\bicsintabular{Ř§ŘąŘŞŮŘ§Řš}{ht}
\bicsintabular{Ř¨Ř˛ŘąÚŻ}{huge}
\bicsintabular{Ř¨Ř˛ŘąÚŻâŘŞŘą}{Huge}
\bicsintabular{Ř§Ř¨ŘąŮžŰŮŮŘŻ}{hyperlink}
\bicsintabular{Ř¨Ř§ŘąÚŻŘ°Ř§ŘąŰâŘ§Ř¨Řą}{hypersetup}
\bicsintabular{ŮŘŻŮâŘ§Ř¨Řą}{hypertarget}
\bicsintabular{ŘŞŰŘąŮâŘ¨ŮŘŻŰ}{hyphenation}
\bicsintabular{ŮŮŰŘłŮâŘŞŰŘąŮ}{hyphenchar}
\bicsintabular{ŘŹŘąŰŮŮâŘŞŰŘąŮâŘ¨ŮŘŻŰ}{hyphenpenalty}
\bicsintabular{@ÚŻŘąÚŠŮŘ§ŘłâŮŘąŘ§ŘŽŮŘ§ŮŰâŘ´ŘŻŮ}{@ifclassloaded}
\bicsintabular{@ÚŻŘąŘŞŘąŘ´ŘŻŮŰ}{@ifdefinable}
\bicsintabular{@ÚŻŘąŮŮŰŘłŮâŘ¨ŘšŘŻŰ}{@ifnextchar}
\bicsintabular{@ÚŻŘąŘłŘ¨ÚŠâŮŘąŘ§ŘŽŮŘ§ŮŰâŘ´ŘŻŮ}{@ifpackageloaded}
\bicsintabular{@ÚŻŘąŘłŘŞŘ§ŘąŮ}{@ifstar}
\bicsintabular{@ÚŻŘąŘŞŘšŘąŰŮâŮŘ´ŘŻŮ}{@ifundefined}
\bicsintabular{ÚŻŘą}{if}
\bicsintabular{ÚŻŘą@ŘłŮŘ§ŮŘŞâŘ˘}{if@tempswa}
\bicsintabular{ÚŻŘąŘ§ŮŮŘ§Řš}{ifcase}
\bicsintabular{ÚŻŘąŘąŘŻŮ}{ifcat}
\bicsintabular{ÚŻŘąŘŞŘšŘąŰŮâŘ´ŘŻŮ}{ifdefined}
\bicsintabular{ÚŻŘąŘ¨ŘšŘŻ}{ifdim}
\bicsintabular{ÚŻŘąŘŞŮâŮžŘąŮŮŘŻŮ}{ifeof}
\bicsintabular{ÚŻŘąŘą}{iff}
\bicsintabular{ÚŻŘąŮŘ§ŘŻŘąŘłŘŞ}{iffalse}
\bicsintabular{ÚŻŘąŮžŘąŮŮŘŻŮâŮŮŘŹŮŘŻ}{IfFileExists}
\bicsintabular{ÚŻŘąÚŠŘ§ŘŻŘąŘ§}{ifhbox}
\bicsintabular{ÚŻŘąŘ­Ř§ŮŘŞâŘ§}{ifhmode}
\bicsintabular{ÚŻŘąŘŻŘąŮŮŰ}{ifinner}
\bicsintabular{ÚŻŘąŘ­Ř§ŮŘŞâŘąŰŘ§ŘśŰ}{ifmmode}
\bicsintabular{ÚŻŘąŘšŘŻŘŻ}{ifnum}
\bicsintabular{ÚŻŘąŮŘąŘŻ}{ifodd}
\bicsintabular{ÚŻŘąŘ˘ŮÚŻŘ§ŮâŘŻŰÚŻŘą}{ifthenelse}
\bicsintabular{ÚŻŘąŘŻŘąŘłŘŞ}{iftrue}
\bicsintabular{ÚŻŘąÚŠŘ§ŘŻŘąŮ}{ifvbox}
\bicsintabular{ÚŻŘąŘ­Ř§ŮŘŞâŮ}{ifvmode}
\bicsintabular{ÚŻŘąŘŞŮŰ}{ifvoid}
\bicsintabular{ÚŻŘąŘŞŘ§Ů}{ifx}
\bicsintabular{ŮŘ§ŘľŮŮâŘŽŘ§ŮŰâŘąŘ§ŮŘŻŰŘŻŮâŘ¨ÚŻŰŘą}{ignorespaces}
\bicsintabular{ŮŮŘąŰ}{immediate}
\bicsintabular{Ř´Ř§ŮŮ}{include}
\bicsintabular{ŘŻŘąŘŹâŘŞŘľŮŰŘą}{includegraphics}
\bicsintabular{ŮŘ´ŮŮŮŰŮ}{includeonly}
\bicsintabular{ŘŞŮŘąŮŘŞÚŻŰ}{indent}
\bicsintabular{ŘŻŘąŮŮŘ§ŰŮ}{index}
\bicsintabular{Ř§ŘłŘŞŘšŮŘ§Ů}{indexentry}
\bicsintabular{ŘšŮŮŘ§ŮâŮŮŘ§ŰŮ}{indexname}
\bicsintabular{ŮŘ§ŘľŮŮâŘąŮŮŮŘ§}{indexspace}
\bicsintabular{ŮŘąŮŘŻŰ}{input}
\bicsintabular{ŮŘąŮŘŻŮžŘąŮŮŘŻŮâÚŻŘąŮŮŘŹŮŘŻ}{InputIfFileExists}
\bicsintabular{Ř´ŮŘ§ŘąŮâŘłŘˇŘąŮŘąŮŘŻŰ}{inputlineno}
\bicsintabular{ŘŻŘąŘŹ}{insert}
\bicsintabular{ŘŹŘąŰŮŮâŘŻŘąŘŹ}{insertpenalties}
\bicsintabular{ŘŹŘąŰŮŮâŘ¨ŰŮâŘłŘˇŘąŮŘ§ŰâŘ˛ŰŘąŮŮŰŘł}{interfootnotelinepenalty}
\bicsintabular{ŘŹŘąŰŮŮâŘ¨ŰŮâŘłŘˇŘąŮŘ§ŰâŮŮŘ§ŰŘ´}{interdisplaylinepenalty}
\bicsintabular{ŘŹŘąŰŮŮâŘ¨ŰŮâŘłŘˇŘąŮŘ§}{interlinepenalty}
\bicsintabular{ŮŘŞŮâŘŻŘ§ŘŽŮŰ}{intertext}
\bicsintabular{ŮŘ§ŘľŮŮâŘ´ŮŘ§ŮŘąŘŻŘąŮŘŞŮ}{intertextsep}
\bicsintabular{ŮŘŽŮŰ}{invisible}
\bicsintabular{ŮžŰŘ´âŮŘąŘśâŘ§Ű}{itdefault}
\bicsintabular{Ř´ÚŠŮâŘ§ŰŘŞŘ§ŮŰÚŠ}{itshape}
\bicsintabular{ŮŮŘąŮ}{item}
\bicsintabular{ŘŞŮŘąŮŘŞÚŻŰâŮŮŘąŮ}{itemindent}
\bicsintabular{ŮŘ§ŘľŮŮâŮŮŘąŮ}{itemsep}
\bicsintabular{ŘŞÚŠŘąŘ§ŘąÚŠŮ}{iterate}
\bicsintabular{Ř´ÚŠŮâŘ§Ű}{itshape}
\bicsintabular{ŮŘ§ŮâÚŠŘ§Řą}{jobname}
\bicsintabular{ŮŮŮž}{jot}
\bicsintabular{ŘŻŮŘąŰ}{kern}
\bicsintabular{Ř§ŮÚŻŮ}{kill}
\bicsintabular{Ř¨ŘąÚŘłŘ¨}{label}
\bicsintabular{Ř¨ŘąÚŘłŘ¨âŘ´ŮŘ§ŘąŘ´âŰÚŠ}{labelenumi}
\bicsintabular{Ř¨ŘąÚŘłŘ¨âŘ´ŮŘ§ŘąŘ´âŘŻŮ}{labelenumii}
\bicsintabular{Ř¨ŘąÚŘłŘ¨âŘ´ŮŘ§ŘąŘ´âŘłŮ}{labelenumiii}
\bicsintabular{Ř¨ŘąÚŘłŘ¨âŘ´ŮŘ§ŘąŘ´âÚŮŘ§Řą}{labelenumiv}
\bicsintabular{Ř¨ŘąÚŘłŘ¨âŮŮŘąŮâŰÚŠ}{labelitemi}
\bicsintabular{Ř¨ŘąÚŘłŘ¨âŮŮŘąŮâŘŻŮ}{labelitemii}
\bicsintabular{Ř¨ŘąÚŘłŘ¨âŮŮŘąŮâŘłŮ}{labelitemiii}
\bicsintabular{Ř¨ŘąÚŘłŘ¨âŮŮŘąŮâÚŮŘ§Řą}{labelitemiv}
\bicsintabular{ŮŘ§ŘľŮŮâŘ§Ř˛Ř¨ŘąÚŘłŘ¨}{labelsep}
\bicsintabular{ŮžŮŮŘ§ŰâŘ¨ŘąÚŘłŘ¨}{labelwidth}
\bicsintabular{Ř˛Ř¨Ř§Ů}{language}
\bicsintabular{ŘŻŘąŘ´ŘŞ}{large}
\bicsintabular{ŘŻŘąŘ´ŘŞâŘŞŘą}{Large}
\bicsintabular{ŘŻŘąŘ´ŘŞâŘŻŘąŘ´ŘŞ}{LARGE}
\bicsintabular{Ř˘ŘŽŘąŰŮâÚŠŘ§ŘŻŘą}{lastbox}
\bicsintabular{Ř˘ŘŽŘąŰŮâŘŻŮŘąŰ}{lastkern}
\bicsintabular{Ř˘ŘŽŘąŰŮâŘŹŘąŰŮŮ}{lastpenalty}
\bicsintabular{Ř˘ŘŽŘąŰŮâŮŘ§ŘľŮŮ}{lastskip}
\bicsintabular{ŮŘ§ŘŞÚŠ}{LaTeX}
\bicsintabular{ŮŘ§ŘŞÚŠâŘ§Ű}{LaTeXe}
\bicsintabular{ÚŠŘŻÚŠŮÚÚŠ}{lccode}
\bicsintabular{ŮŮŘ§ŘˇâŘŽ}{ldots}
\bicsintabular{ŮŘ´Ř§ŮÚŻŘą}{leaders}
\bicsintabular{ŘŞŘąÚŠâŮ}{leavevmode}
\bicsintabular{ÚŮž}{left}
\bicsintabular{Ř­Ř§Ř´ŰŮâÚŮž}{leftmargin}
\bicsintabular{Ř­Ř§Ř´ŰŮâÚŮžâŰÚŠ}{leftmargini}
\bicsintabular{Ř­Ř§Ř´ŰŮâÚŮžâŘŻŮ}{leftmarginii}
\bicsintabular{Ř­Ř§Ř´ŰŮâÚŮžâŘłŮ}{leftmarginiii}
\bicsintabular{Ř­Ř§Ř´ŰŮâÚŮžâÚŮŘ§Řą}{leftmarginiv}
\bicsintabular{Ř­Ř§Ř´ŰŮâÚŮžâŮžŮŘŹ}{leftmarginv}
\bicsintabular{Ř­Ř§Ř´ŰŮâÚŮžâŘ´Ř´}{leftmarginvi}
\bicsintabular{ŘšŮŘ§ŮŘŞâÚŮž}{leftmark}
\bicsintabular{ÚŠŘ§ŘŻŘąŘŞŘ§ÚŮž}{leftpageskip}
\bicsintabular{ŮŘ§ŘľŮŮâŘ§Ř¨ŘŞŘŻŘ§ŰâŘłŘˇŘą}{leftskip}
\bicsintabular{Ř¨ÚŻŘ°Ř§Řą}{let}
\bicsintabular{ŘłŘˇŘą}{line}
\bicsintabular{ŘłŘˇŘąŘ´ÚŠŮ}{linebreak}
\bicsintabular{ŘŹŘąŰŮŮâŘłŘˇŘą}{linepenalty}
\bicsintabular{ŮŘ§ŘľŮŮâŘłŘˇŘąŮŘ§}{lineskip}
\bicsintabular{Ř­ŘŻŮŘ§ŘľŮŮâŘłŘˇŘą}{lineskiplimit}
\bicsintabular{ÚŠŘ´Ř´âŮŘ§ŘľŮŮâŘłŘˇŘą}{linespread}
\bicsintabular{ŘśŘŽŘ§ŮŘŞâŘŽŘˇ}{linethickness}
\bicsintabular{ŮžŮŮŘ§ŰâŘłŘˇŘą}{linewidth}
\bicsintabular{ŘšŮŮŘ§ŮâŮŮŘąŘłŘŞâŘ§Ř´ÚŠŘ§Ů}{listfigurename}
\bicsintabular{ŮŰŘłŘŞâŮžŘąŮŮŘŻŮâŮŘ§}{listfiles}
\bicsintabular{ŮŮŘąŘłŘŞâŘ§Ř´ÚŠŘ§Ů}{listoffigures}
\bicsintabular{ŮŮŘąŘłŘŞâŘŹŘŻŘ§ŮŮ}{listoftables}
\bicsintabular{ŘŞŮŘąŮŘŞÚŻŰâŘ¨ŮŘŻŮŰŘłŘŞ}{listparindent}
\bicsintabular{ŘšŮŮŘ§ŮâŮŮŘąŘłŘŞâŘŹŘŻŘ§ŮŮ}{listtablename}
\bicsintabular{Ř¨Ř§ŘąÚŠŮâŘˇŘ¨ŮŮ}{LoadClass}
\bicsintabular{Ř¨Ř§ŘąÚŠŮâŘˇŘ¨ŮŮâŘ¨Ř§ÚŻŘ˛ŰŮŮ}{LoadClassWithOptions}
\bicsintabular{ŮÚŠŘ§Ů}{location}
\bicsintabular{Ř¨ŮŮŘŻ}{long}
\bicsintabular{Ř­ŮŮŮ}{loop}
\bicsintabular{ÚŻŘłŰŘŽŘŞÚŻŰ}{looseness}
\bicsintabular{Ř§ŮŘŞŮŘ§ŮâŘ¨ŮžŘ§ŰŰŮ}{lower}
\bicsintabular{@ŘŻŰÚŻŘą}{@makeother}
\bicsintabular{@Ř˛Ř§Řą}{@m}
\bicsintabular{ŘŻŮ@Ř˛Ř§Řą}{@M}
\bicsintabular{ŘŻŮ@Ř˛Ř§ŘąŰÚŠ}{@Mi}
\bicsintabular{ŘŻŮ@Ř˛Ř§ŘąŘŻŮ}{@Mii}
\bicsintabular{ŘŻŮ@Ř˛Ř§ŘąŘłŮ}{@Miii}
\bicsintabular{ŘŻŮ@Ř˛Ř§ŘąÚŮŘ§Řą}{@Miv}
\bicsintabular{Ř¨ŰŘł@Ř˛Ř§Řą}{@MM}
\bicsintabular{ŮŮ@Ř§}{m@ne}
\bicsintabular{Ř¨Ř˛ŘąÚŻâŮŮŘ§ŰŰ}{mag}
\bicsintabular{ÚŻŘ§Ů}{magstep}
\bicsintabular{ŮŰŮâÚŻŘ§Ů}{magstephalf}
\bicsintabular{ŮŘˇŮŘ¨âŘ§ŘľŮŰ}{mainmatter}
\bicsintabular{Ř§ŘŞâŘ­ŘąŮ}{makeatletter}
\bicsintabular{Ř§ŘŞâŘŻŰÚŻŘą}{makeatother}
\bicsintabular{ÚŠŘ§ŘŻŘąŘ¨ŰâŘŽŘˇ}{makebox}
\bicsintabular{ŘłŘ§ŘŽŘŞâŮŘąŮŮÚŻ}{makeglossary}
\bicsintabular{ŘŞŮŰŮâŮŮŘ§ŰŮ}{makeindex}
\bicsintabular{ŘłŘ§ŘŽŘŞâŘ¨ŘąÚŘłŘ¨}{makelabel}
\bicsintabular{ŘłŘ§ŘŽŘŞâŘ¨ŘąÚŘłŘ¨âŮŘ§}{makelabels}
\bicsintabular{ŘłŘ§ŘŽŘŞâŘ­ŘąŮŮâÚŠŮÚÚŠ}{MakeLowercase}
\bicsintabular{ŘšŮŮŘ§ŮâŘłŘ§Ř˛}{maketitle}
\bicsintabular{ŘłŘ§ŘŽŘŞâŘ­ŘąŮŮâŘ¨Ř˛ŘąÚŻ}{MakeUppercase}
\bicsintabular{ŘŻŘąŘ­Ř§Ř´ŰŮ}{marginpar}
\bicsintabular{ŮŘ§ŘľŮŮâŘŻŮŘ­Ř§Ř´ŰŮ}{marginparpush}
\bicsintabular{ŮŘ§ŘľŮŮâŘŞŘ§Ř­Ř§Ř´ŰŮ}{marginparsep}
\bicsintabular{ŮžŮŮŘ§ŰâŘ­Ř§Ř´ŰŮ}{marginparwidth}
\bicsintabular{ŘšŮŘ§ŮŘŞ}{mark}
\bicsintabular{ŘšŮŘ§ŮŘŞâŘŻŘąŘŻŮŘˇŘąŮ}{markboth}
\bicsintabular{ŘšŮŘ§ŮŘŞâŘŻŘąŘąŘ§ŘłŘŞ}{markright}
\bicsintabular{Ř§ŘšŘąŘ§Ř¨âŘąŰŘ§ŘśŰ}{mathaccent}
\bicsintabular{ŮŮŰŘłŮâŘąŰŘ§ŘśŰ}{mathchar}
\bicsintabular{ŘŞŘšŘąŰŮâŮŮŰŘłŮâŘąŰŘ§ŘśŰ}{mathchardef}
\bicsintabular{ÚŠŘŻŘąŰŘ§ŘśŰ}{mathcode}
\bicsintabular{ŘąŰŘ§ŘśŰâŘąŮŮŮ}{mathrm}
\bicsintabular{Ř­ŘŻŘ§ÚŠŘŤŘąŘŞÚŠŘąŘ§Řą}{maxdeadcycles}
\bicsintabular{Ř­ŘŻŘ§ÚŠŘŤŘąŘšŮŮâŘľŮŘ­Ů}{maxdepth}
\bicsintabular{Ř¨ŘšŘŻŘ¨ŰŘ´ŰŮ}{maxdimen}
\bicsintabular{ÚŠŘ§ŘŻŘąŘ¨Ű}{mbox}
\bicsintabular{Ř´ŮŘ§ŰŮâŮŘ§Ř˛ÚŠ}{mdseries}
\bicsintabular{ŮŘšŮŘ§}{meaning}
\bicsintabular{ŮŘ§Ř˛ÚŠ}{mediumseries}
\bicsintabular{ŮŘ§ŘľŮŮâŮŘŞŮŘłŘˇâŘąŰŘ§ŘśŰ}{medmuskip}
\bicsintabular{ŮžŘąŘ´âŮŘŞŮŘłŘˇ}{medskip}
\bicsintabular{ŮŮŘŻŘ§ŘąŮžŘąŘ´âŮŘŞŮŘłŘˇ}{medskipamount}
\bicsintabular{ŮŘśŘ§ŰâŮŘŞŮŘłŘˇ}{medspace}
\bicsintabular{ŮžŰŘ§Ů}{message}
\bicsintabular{ŮžŰŘ§ŮâŘ´ÚŠŮ}{MessageBreak}
\bicsintabular{Ř­ŘŻŘ§ŮŮâŮŘ§ŘľŮŮâŘąŘŻŰŮ}{minrowclearance}
\bicsintabular{ŘŻŮŘąŰâŘąŰŘ§ŘśŰ}{mkern}
\bicsintabular{ŮŘ§Ů}{month}
\bicsintabular{Ř§ŮŘŞŮŘ§ŮâŘ¨ÚŮž}{moveleft}
\bicsintabular{Ř§ŮŘŞŮŘ§ŮâŘ¨ŘąŘ§ŘłŘŞ}{moveright}
\bicsintabular{ŮŘ§ŘľŮŮâŘąŰŘ§ŘśŰ}{mskip}
\bicsintabular{ŘąŰ@ŘśŰ}{m@th}
\bicsintabular{ÚŮŘŻŘłŘŞŮŮŰ}{multicolumn}
\bicsintabular{ŘśŘąŘ¨}{multiply}
\bicsintabular{ÚŮŘŻŘ§ŘŻŘşŘ§Ů}{multispan}
\bicsintabular{ŮŰŮŮŘ§ŘľŮŮ}{muskip}
\bicsintabular{ŘŞŘšŘąŰŮâŮŰŮŮŘ§ŘľŮŮ}{muskipdef}
\bicsintabular{@ŘŞŘąŮŘ§Ů}{@namedef}
\bicsintabular{@ÚŠŘ§ŘąŘ¨ŘąŘŻŮŘ§Ů}{@nameuse}
\bicsintabular{ŰÚŠ@}{@ne}
\bicsintabular{ŮŘ§Ů}{name}
\bicsintabular{ŘˇŘ¨ŰŘšŰ}{natural}
\bicsintabular{Ř¨Ř§ŘąŰÚŠ}{nearrow}
\bicsintabular{Ř¨Ř§ŘąŰÚŠŘŞŘą}{nearrower}
\bicsintabular{Ř´ÚŠŮŘ¨ŮŘŻŰâŮŮŘąŘŻŮŰŘ§Ř˛}{NeedsTeXFormat}
\bicsintabular{ŮŮŮŰ}{neg}
\bicsintabular{ŮŘśŘ§ŰâŮŘŞŮŘłŘˇâŮŮŮŰ}{negmedspace}
\bicsintabular{ŮŘśŘ§ŰâŘśŘŽŰŮâŮŮŮŰ}{negthickspace}
\bicsintabular{ŘŻŮŘąŰâÚŠŮÚÚŠâŮŮŮŰ}{negthinspace}
\bicsintabular{Ř¨ŮŮŰâŮŮ}{newboolean}
\bicsintabular{ÚŠŘ§ŘŻŘąŘŹŘŻŰŘŻ}{newbox}
\bicsintabular{ŮŘąŮŘ§ŮâŮŮ}{newcommand}
\bicsintabular{Ř´ŮŘ§ŘąŘŹŘŻŰŘŻ}{newcount}
\bicsintabular{Ř´ŮŘ§ŘąŮŘŻŮâŘŹŘŻŰŘŻ}{newcounter}
\bicsintabular{Ř¨ŘšŘŻŘŹŘŻŰŘŻ}{newdimen}
\bicsintabular{ŮŘ­ŰŘˇâŮŮ}{newenvironment}
\bicsintabular{ŘŽŘ§ŮŮŘ§ŘŻŮâŘŹŘŻŰŘŻ}{newfam}
\bicsintabular{ŮŮŮâŮŮ}{newfont}
\bicsintabular{ÚŠŮÚŠâŘŹŘŻŰŘŻ}{newhelp}
\bicsintabular{ÚŻŘąŘŹŘŻŰŘŻ}{newif}
\bicsintabular{ŘŻŘąŘŹâŘŹŘŻŰŘŻ}{newinsert}
\bicsintabular{Ř¨ŘąÚŘłŘ¨âŘŹŘŻŰŘŻ}{newlabel}
\bicsintabular{ŘŞŘšŘąŰŮâŘ¨ŘšŘŻŘŹŘŻŰŘŻ}{newlength}
\bicsintabular{ŘłŘˇŘąŘŹŘŻŰŘŻ}{newline}
\bicsintabular{ŮŮŰŘłŮâŘłŘˇŘąŘŹŘŻŰŘŻ}{newlinechar}
\bicsintabular{ŮŰŮŮŘ§ŘľŮŮâŘŹŘŻŰŘŻ}{newmuskip}
\bicsintabular{ŘľŮŘ­ŮâŘŹŘŻŰŘŻ}{newpage}
\bicsintabular{Ř¨ŘŽŮŘ§ŮâŘŹŘŻŰŘŻ}{newread}
\bicsintabular{ŘŞŘšŘąŰŮâÚŠŘ§ŘŻŘąŘŹŘŻŰŘŻ}{newsavebox}
\bicsintabular{ŮŘ§ŘľŮŮâŘŹŘŻŰŘŻ}{newskip}
\bicsintabular{ŮŘśŰŮâŘŹŘŻŰŘŻ}{newtheorem}
\bicsintabular{ŘŹŘ˛ŘĄŘŹŘŻŰŘŻ}{newtoks}
\bicsintabular{Ř¨ŮŮŰŘłâŘŹŘŻŰŘŻ}{newwrite}
\bicsintabular{Ř¨ŰâŘąŘŻŰŮ}{noalign}
\bicsintabular{ŮŘ´ÚŠŮ}{nobreak}
\bicsintabular{ŮŘ§ŘľŮŮâŮŘ´ÚŠŘłŘŞŮŰ}{nobreakspace}
\bicsintabular{Ř¨ŘŻŮŮâŘłŮŘŻ}{nocite}
\bicsintabular{ŮÚŻŘłŘŞŘą}{noexpand}
\bicsintabular{Ř¨ŘŻŮŮâŮžŘąŮŮŘŻŮ}{nofiles}
\bicsintabular{Ř¨ŘŻŮŮâŘŞŮŘąŮŘŞÚŻŰ}{noindent}
\bicsintabular{Ř¨ŰâŮŘ§ŘľŮŮâŘłŘˇŘą}{nointerlineskip}
\bicsintabular{Ř¨ŘŻŮŮâŘ­ŘŻ}{nolimits}
\bicsintabular{ŘłŘˇŘąŮŘ´ÚŠŮ}{nolinebreak}
\bicsintabular{ŮžŘąŘŻŘ§Ř˛Ř´âŘ¨ŘŻŮŮâŘŞŮŮŮ}{nonstopmode}
\bicsintabular{ŮŮŘ§ŘľŮâŮŘŞŘšŘ§ŘąŮâŮŘ§ŘŞŰŮ}{nonfrenchspacing}
\bicsintabular{Ř¨ŘŻŮŮâŘ´ŮŘ§ŘąŮ}{nonumber}
\bicsintabular{ŘľŮŘ­ŮâŮŘ´ÚŠŮ}{nopagebreak}
\bicsintabular{ÚŠŘąŘłŰŮŘ§ŰâŮŘŞŘšŘ§ŘąŮ}{normalbaselines}
\bicsintabular{ŮŘ§ŘľŮŮâÚŠŘąŘłŰâŮŘŞŘšŘ§ŘąŮ}{normalbaselineskip}
\bicsintabular{ŘąŮÚŻâŘšŘ§ŘŻŰ}{normalcolor}
\bicsintabular{ŮŮŮâŘšŘ§ŘŻŰ}{normalfont}
\bicsintabular{ŮŘ§ŘľŮŮâŘłŘˇŘąŮŘŞŘšŘ§ŘąŮ}{normallineskip}
\bicsintabular{Ř­ŘŻŮŘ§ŘľŮŮâŘłŘˇŘąŮŘŞŘšŘ§ŘąŮ}{normallineskiplimit}
\bicsintabular{ŘŻŘąŘ­Ř§Ř´ŰŮâŘšŘ§ŘŻŰ}{normalmarginpar}
\bicsintabular{Ř§ŮŘŻŘ§Ř˛ŮâŘšŘ§ŘŻŰ}{normalsize}
\bicsintabular{Ř¨ŘŻŮŮâŘ§ŘŞŰÚŠŘŞ}{notag}
\bicsintabular{ŮŮŮ}{null}
\bicsintabular{ŮŮŮâŘŞŮŰ}{nullfont}
\bicsintabular{ŘšŘŻŘŻ}{number}
\bicsintabular{ŘłŘˇŘąŘšŘŻŘŻŰ}{numberline}
\bicsintabular{Ř´ŮŘ§ŘąŮâŮŘˇŘ§Ř¨Ů}{numberwithin}
\bicsintabular{ŮžŘ§ŰŰŮâŘľŮŘ­ŮâŘ˛ŮŘŹ}{@evenfoot}
\bicsintabular{Ř¨Ř§ŮŘ§ŰâŘľŮŘ­ŮâŘ˛ŮŘŹ}{@evenhead}
\bicsintabular{ŮžŘ§ŰŰŮâŘľŮŘ­ŮâŮŘąŘŻ}{@oddfoot}
\bicsintabular{Ř¨Ř§ŮŘ§ŰâŘľŮŘ­ŮâŮŘąŘŻ}{@oddhead}
\bicsintabular{Ř´ŮŘ§ŘąŮâŘ¨ŰŘąŮŮâŘŻŘąŘłŘŞ}{@outeqntrue}
\bicsintabular{Ř´ŮŘ§ŘąŮâŘ¨ŰŘąŮŮâŮŘ§ŘŻŘąŘłŘŞ}{@outeqnfalse}
\bicsintabular{ŘłŘˇŘąŘ¨ŮâŘłŘˇŘą}{obeylines}
\bicsintabular{ŮŘśŘ§ŮŘšŘ§Ů}{obeyspaces}
\bicsintabular{Ř­Ř§Ř´ŰŮâŮŘąŘŻ}{oddsidemargin}
\bicsintabular{ŘłŘˇŮŘąŘ¨ŰâŮŘ§ŘľŮŮ}{offinterlineskip}
\bicsintabular{Ř­Ř°Ů}{omit}
\bicsintabular{@ŘŞŮŮŘ§ŘŻŘąŮžŰŘ´âŘŻŘąŘ˘ŮŘŻ}{@onlypreamble}
\bicsintabular{ŰÚŠâŘłŘŞŮŮ}{onecolumn}
\bicsintabular{ŘŞŮŮŘ§âŰŘ§ŘŻŘŻŘ§Ř´ŘŞâŮŘ§}{onlynotes}
\bicsintabular{ŘŞŮŮŘ§Ř§ŘłŮŘ§ŰŘŻŮŘ§}{onlyslides}
\bicsintabular{Ř¨Ř§Ř˛ÚŠŮâŮŘąŮŘŻŰ}{openin}
\bicsintabular{Ř¨Ř§Ř˛ÚŠŮâŘŽŘąŮŘŹŰ}{openout}
\bicsintabular{ÚŻŘ˛ŰŮŮâŮŘľŘąŮâŮŘ´ŘŻŮ}{OptionNotUsed}
\bicsintabular{ŰŘ§}{or}
\bicsintabular{Ř¨ŘąŮŮŰ}{outer}
\bicsintabular{ŘľŮŘ­ŮâŘ¨ŮŘŻŰ}{output}
\bicsintabular{ŘŹŘąŰŮŮâŘľŮŘ­ŮâŘ¨ŮŘŻŰ}{outputpenalty}
\bicsintabular{ŘšŮŘ§ŮŘŞâŘłŘąŘąŰŘ˛}{overfullrule}
\bicsintabular{@ŮŘąŮŘ§ŮâŮŘ§ŰâŮžŰŘ´âŘŻŘąŘ˘ŮŘŻ}{@preamblecmds}
\bicsintabular{@ŮžŮ}{p@}
\bicsintabular{ŘŽŘˇŘ§ŰâŘłŘ¨ÚŠ}{PackageError}
\bicsintabular{Ř§ŘˇŮŘ§ŘšâŘłŘ¨ÚŠ}{PackageInfo}
\bicsintabular{ŮŘ´ŘŻŘ§ŘąŘłŘ¨ÚŠ}{PackageWarning}
\bicsintabular{ŮŘ´ŘŻŘ§ŘąŘłŘ¨ÚŠâŘ¨ŰâŘłŘˇŘą}{PackageWarningNoLine}
\bicsintabular{ŘľŮŘ­ŮâŘ´ÚŠŮ}{pagebreak}
\bicsintabular{ŘąŮÚŻâŘľŮŘ­Ů}{pagecolor}
\bicsintabular{ŘšŮŮâŘľŮŘ­Ů}{pagedepth}
\bicsintabular{ÚŠŘ´Ř´âŮžŘąŘąŘąŘľŮŘ­Ů}{pagefilllstretch}
\bicsintabular{ÚŠŘ´Ř´âŮžŘąŘąŘľŮŘ­Ů}{pagefillstretch}
\bicsintabular{ÚŠŘ´Ř´âŮžŘąŘľŮŘ­Ů}{pagefilstretch}
\bicsintabular{ŘşŘ§ŰŘŞâŘľŮŘ­Ů}{pagegoal}
\bicsintabular{ŮŘ§ŮâŘľŮŘ­Ů}{pagename}
\bicsintabular{Ř´ŮŘ§ŘąŮâÚŻŘ°Ř§ŘąŰâŘľŮŘ­Ů}{pagenumbering}
\bicsintabular{ŘąŘŹŮŘšâŘľŮŘ­Ů}{pageref}
\bicsintabular{ŘśŘŽŘ§ŮŘŞâŘŽŘˇâŘľŮŘ­Ů}{pagerulewidth}
\bicsintabular{ŮŘ´ŘąŘŻÚŻŰâŘľŮŘ­Ů}{pageshrink}
\bicsintabular{ÚŠŘ´Ř´âŘľŮŘ­Ů}{pagestretch}
\bicsintabular{ŘłŘ¨ÚŠâŘľŮŘ­Ů}{pagestyle}
\bicsintabular{ŘŹŮŘšâŘľŮŘ­Ů}{pagetotal}
\bicsintabular{Ř¨ŮŮŘŻŘ§ŰâÚŠŘ§ŘşŘ°}{paperheight}
\bicsintabular{ŮžŮŮŘ§ŰâÚŠŘ§ŘşŘ°}{paperwidth}
\bicsintabular{Ř¨ŮŘŻ}{par}
\bicsintabular{ŮžŘ§ŘąŘ§ÚŻŘąŘ§Ů}{paragraph}
\bicsintabular{ŮŮŘ§Ř˛Ű}{parallel}
\bicsintabular{ÚŠŘ§ŘŻŘąŮžŘ§Řą}{parbox}
\bicsintabular{ŮŘ§ŘľŮŮâŘŞŮâŘ¨ŮŘŻ}{parfillskip}
\bicsintabular{ŘŞŮŘąŮŘŞÚŻŰâŘłŘąŘ¨ŮŘŻ}{parindent}
\bicsintabular{ŮŘ§ŘľŮŮâŘ¨ŮŘŻŮŰŘłŘŞ}{parsep}
\bicsintabular{Ř´ÚŠŮâŘ¨ŮŘŻ}{parshape}
\bicsintabular{ŮŘ§ŘľŮŮâŘ¨ŮŘŻ}{parskip}
\bicsintabular{Ř¨ŘŽŘ´}{part}
\bicsintabular{ŘšŮŮŘ§ŮâŘ¨ŘŽŘ´}{partname}
\bicsintabular{ŮŘ§ŘľŮŮâŘ¨Ř§ŮŘ§ŰâŮŰŘłŘŞâŘ¨ŮŘŻ}{partopsep}
\bicsintabular{Ř§ŘąŘłŘ§ŮâÚŻŘ˛ŰŮŮâŘ¨ŮâÚŠŮŘ§Řł}{PassOptionToClass}
\bicsintabular{Ř§ŘąŘłŘ§ŮâÚŻŘ˛ŰŮŮâŘ¨ŮâŮžÚŠŰŘŹ}{PassOptionToPackage}
\bicsintabular{ŮŘłŰŘą}{path}
\bicsintabular{Ř§ŮÚŻŮŮŘ§}{patterns}
\bicsintabular{ŮÚŠŘŤ}{pausing}
\bicsintabular{ŘŹŘąŰŮŮ}{penalty}
\bicsintabular{ŘşŰŘ¨}{phantom}
\bicsintabular{Ř§ŮÚŻŮŰâŮŘ¨ŮŰ}{poptabs}
\bicsintabular{ŘŹŘąŰŮŮâŮžŘłâŮŮŘ§ŰŘ´}{postdisplaypenalty}
\bicsintabular{ŘŹŮŘŞâŮžŰŘ´âŮŮŘ§ŰŘ´}{predisplaydirection}
\bicsintabular{ŘŹŘąŰŮŮâŮžŰŘ´âŮŮŘ§ŰŘ´}{predisplaypenalty}
\bicsintabular{Ř§ŮŘŻŘ§Ř˛ŮâŮžŰŘ´âŮŮŘ§ŰŘ´}{predisplaysize}
\bicsintabular{ŮžŰŘ´âŘ­ŘŻŘ¨ŘŻŮŮŘ§ŰŰ}{pretolerance}
\bicsintabular{ŘšŮŮâŮŘ¨ŮŰ}{prevdepth}
\bicsintabular{Ř¨ŮŘŻŮŘ¨ŮŰ}{prevgraf}
\bicsintabular{ŮŮŘ§ŰŮâŘŻŘąŘ§ŰŮŘŹŘ§}{printindex}
\bicsintabular{ŮžŘąŘŻŘ§Ř˛Ř´âÚŻŘ˛ŰŮŮâŮŘ§}{ProcessOptions}
\bicsintabular{ŘŞŘ§ŮŰŮ}{protect}
\bicsintabular{ŘŞŮŰŮâŮŘąŮŘ§Ů}{providecommand}
\bicsintabular{Ř˘ŮŘ§ŘŻŮâŘłŘ§Ř˛ŰâŘˇŘ¨ŮŮ}{ProvidesClass}
\bicsintabular{Ř˘ŮŘ§ŘŻŮâŘłŘ§Ř˛ŰâŮžŘąŮŮŘŻŮ}{ProvidesFile}
\bicsintabular{Ř˘ŮŘ§ŘŻŮâŘłŘ§Ř˛ŰâŘłŘ¨ÚŠ}{ProvidesPackage}
\bicsintabular{ŘŤŘ¨ŘŞâŘ§ŮÚŻŮ}{pushtabs}
\bicsintabular{ÚŠŮŘ§ŘŻ}{quad}
\bicsintabular{ÚŠŮÚŠŮŘ§ŘŻ}{qquad}
\bicsintabular{@Ř¨Ř§Ř˛Ř˘ŰŰâŘŽŘąŮŘŹâŘľŮŘ­Ů}{@outputpagerestore}
\bicsintabular{ŘąŘ§ŘŻŰÚŠŘ§Ů}{radical}
\bicsintabular{ŮžŘ§ŰŰŮâŘ¨ŰâŘŞŮŘ¸ŰŮ}{raggedbottom}
\bicsintabular{ŘŞŮŘ¸ŰŮâŘ§Ř˛ŘąŘ§ŘłŘŞ}{raggedleft}
\bicsintabular{ŘŞŮŘ¸ŰŮâŘ§Ř˛ÚŮž}{raggedright}
\bicsintabular{Ř§ŮŘŞŮŘ§ŮâŘ¨Ř¨Ř§ŮŘ§}{raise}
\bicsintabular{Ř¨Ř§ŮŘ§Ř¨Řą}{raisebox}
\bicsintabular{ŘŞŘąŮŰŘšâŘ§ŘŞŰÚŠŘŞ}{raisetag}
\bicsintabular{Ř˛Ř§ŮŰŮâŘą}{rangle}
\bicsintabular{ŘłŮŮâŘą}{rceil}
\bicsintabular{Ř¨ŘŽŮŘ§Ů}{read}
\bicsintabular{ŘąŘŹŮŘš}{ref}
\bicsintabular{ÚŠŘ§ŘŻŘąŮŘąŰŮŮ}{reflectbox}
\bicsintabular{ŘšŮŮŘ§ŮâŮŘąŘ§ŘŹŘš}{refname}
\bicsintabular{ÚŻŘ§ŮâŘ´ŮŘ§ŘąŮŘŻŮâŮŘąŘŹŘš}{refstepcounter}
\bicsintabular{ŘąŘ§Ř­ŘŞ}{relax}
\bicsintabular{ŘąŮŘšâŘ˘ŘŽŘąŰŮâŮŘ§ŘľŮŮ}{removelastskip}
\bicsintabular{ŮŘąŮŘ§ŮâŘ§Ř˛ŮŮ}{renewcommand}
\bicsintabular{ŮŘ­ŰŘˇâŘ§Ř˛ŮŮ}{renewenvironment}
\bicsintabular{Ř§Ř˛ŮŮ}{repeat}
\bicsintabular{ŘłŘ¨ÚŠâŮŮŘąŘŻŮŰŘ§Ř˛}{RequirePackage}
\bicsintabular{ŘłŘ¨ÚŠâŮŮŘąŘŻŮŰŘ§Ř˛Ř¨Ř§ÚŻŘ˛ŰŮŮ}{RequirePackageWithOptions}
\bicsintabular{ÚŠŘ§ŘŻŘąÚŠŘ´ŰŘŻŮ}{resizebox}
\bicsintabular{ŘŻŘąŘ­Ř§Ř´ŰŮâŮŘšÚŠŮŘł}{reversemarginpar}
\bicsintabular{ÚŠŮâŘą}{rfloor}
\bicsintabular{ŘąŘ§ŘłŘŞ}{right}
\bicsintabular{Ř­Ř§Ř´ŰŮâŘąŘ§ŘłŘŞ}{rightmargin}
\bicsintabular{ŘšŮŘ§ŮŘŞâŘąŘ§ŘłŘŞ}{rightmark}
\bicsintabular{ÚŠŘ§ŘŻŘąŘŞŘ§ŘąŘ§ŘłŘŞ}{rightpageskip}
\bicsintabular{ŮŘ§ŘľŮŮâŘ§ŮŘŞŮŘ§ŰâŘłŘˇŘą}{rightskip}
\bicsintabular{ŘąŮŮŮâŘšŘ§ŘŻŰ}{rmdefault}
\bicsintabular{ŮŘ§ŮŰŮâŘąŮŮŮ}{rmfamily}
\bicsintabular{ŘąŮŮŮâŘ¨Ř˛ŘąÚŻ}{Roman}
\bicsintabular{ŘąŮŮŮâÚŠŮÚÚŠ}{roman}
\bicsintabular{ŘšŘŻŘŻŘąŮŮŰ}{romannumeral}
\bicsintabular{ÚŠŘ§ŘŻŘąÚŘąŘŽŘ§Ů}{rotatebox}
\bicsintabular{ŘąŮÚŻâŘąŘŻŰŮ}{rowcolor}
\bicsintabular{ŘŽŘˇ}{rule}
\bicsintabular{@ŘŻŮŮŰâŘ§Ř˛ŘŻŮ}{@secondoftwo}
\bicsintabular{@ŮŘśŘ§ŮŘ§}{@spaces}
\bicsintabular{ŮŮŰŮâŘľŮŘ­Ů}{samepage}
\bicsintabular{ŮŮŘŻŘ§ŘąÚŠŘ§ŘŻŘą}{savebox}
\bicsintabular{ŮŮÚŠŘ§ŘŻŘą}{sbox}
\bicsintabular{ÚŠŘ§ŘŻŘąŘ§ŮŘŻŘ§Ř˛Ů}{scalebox}
\bicsintabular{ŮžŰŘ´âŮŘąŘśâŘŞŮŘ§ŮâŘ¨Ř˛ŘąÚŻ}{scdefault}
\bicsintabular{Ř´ÚŠŮâŘŞŮŘ§ŮâŘ¨Ř˛ŘąÚŻ}{scshape}
\bicsintabular{ŮŮŮâŘŞŮŘ§Ů}{scriptfont}
\bicsintabular{ŮŮŮâŘŞŮŘ§ŮâŘŞŮŘ§Ů}{scriptscriptfont}
\bicsintabular{ŘłŘ¨ÚŠâŘŞŮâŮŮŘ´ŘŞâŘŞŮâŮŮŘ´ŘŞ}{scriptscriptstyle}
\bicsintabular{Ř§ŮŘŻŘ§Ř˛ŮâŮžŘ§ŮŮŰŘł}{scriptsize}
\bicsintabular{ŘłŘ¨ÚŠâŘŞŮâŮŮŘ´ŘŞ}{scripstyle}
\bicsintabular{ŮžŘąŘŻŘ§Ř˛Ř´âÚŻŘ°ŘąŰ}{scrollmode}
\bicsintabular{ŮŘłŮŘŞ}{section}
\bicsintabular{ŘŞŘšŘąŰŮâŮŘłŮŘŞ}{secdef}
\bicsintabular{Ř¨Ř¨ŰŮŰŘŻ}{see}
\bicsintabular{ŮŰŘ˛Ř¨Ř¨ŰŮŰŘŻ}{seealso}
\bicsintabular{ŮŘ§ŮâŘ¨Ř¨ŰŮŰŘŻ}{seename}
\bicsintabular{ŮŮŮâŘ¨ŘąŘŻŘ§Řą}{selectfont}
\bicsintabular{ŘŞŮŘ¸ŰŮâŘ¨ŮŮŰ}{setboolean}
\bicsintabular{ŘŻŘąÚŠŘ§ŘŻŘą}{setbox}
\bicsintabular{ŮŮŘŻŘ§ŘąŘ´ŮŘ§ŘąŮŘŻŮ}{setcounter}
\bicsintabular{ŮŮŘŻŘ§ŘąÚŠŮŰŘŻŮŘ§}{setkeys}
\bicsintabular{ŮŮŘŻŘ§ŘąŘ¨ŘšŘŻ}{setlength}
\bicsintabular{ŘŞŮŘ¸ŰŮâŮŮŮŘ§}{setminus}
\bicsintabular{ŘŞŘšŘąŰŮâŮŮŮâŘšŮŘ§ŘŚŮ}{SetSymbolFont}
\bicsintabular{ŘŞŮŘ¸ŰŮâŘ¨ŮâŘšŮŮ}{settodepth}
\bicsintabular{ŘŞŮŘ¸ŰŮâŘ¨ŮâŘ§ŘąŘŞŮŘ§Řš}{settoheight}
\bicsintabular{ŮŮŘŻŘ§ŘąŘ¨ŘšŘŻŘ¨ŮâŘ§ŮŘŻŘ§Ř˛Ů}{settowidth}
\bicsintabular{ÚŠŘŻŘśŘąŰŘ¨âŮŘ§ŘľŮŮ}{sfcode}
\bicsintabular{ŮžŰŘ´âŮŘąŘśâŘłâŮ}{sfdefault}
\bicsintabular{ŮŘ§ŮŰŮâŘłŮâŘłŘąŰŮ}{sffamily}
\bicsintabular{ÚŠŘ§ŘŻŘąŘłŘ§ŰŮâŘŻŘ§Řą}{shadowbox}
\bicsintabular{ŘŞŰŘ˛}{sharp}
\bicsintabular{Ř¨ŮŘąŘłŘŞ}{shipout}
\bicsintabular{ŮžŘ´ŘŞŮâÚŠŮŘŞŘ§Ů}{shortstack}
\bicsintabular{ŮŮŘ§ŰŘ´âŘ¨ŘŻŮ}{show}
\bicsintabular{ŮŮŘ§ŰŘ´âŘ¨ŘŻŮâÚŠŘ§ŘŻŘą}{showbox}
\bicsintabular{ŮŰŘ˛Ř§ŮâŮŮŘ§ŰŘ´âÚŠŘ§ŘŻŘą}{showboxbreadth}
\bicsintabular{ŘšŮŮâŮŮŘ§ŰŘ´âÚŠŘ§ŘŻŘą}{showboxdepth}
\bicsintabular{ŮŮŘ§ŰŘ´âŘ¨ŘŻŮâŮŰŘłŘŞŮŘ§}{showlists}
\bicsintabular{ŮŮŘ§ŰŘ´âŘ¨ŘŻŮâŮŘ­ŘŞŮŘ§Ű}{showthe}
\bicsintabular{Ř­Ř§ŮŘŞâŘłŘ§ŘŻŮâŮŮŮ}{simplefontmode}
\bicsintabular{Ř´Ř§ŮŘ˛ŘŻ@}{sixt@@n}
\bicsintabular{ŮŮŰŘłŮâŘ§ŘąŰŘ¨}{skewchar}
\bicsintabular{ŮŘ§ŘľŮŮ}{skip}
\bicsintabular{ŮŘ§ŘľŮ@}{skip@}
\bicsintabular{ŘŞŘšŘąŰŮâŮŘ§ŘľŮŮ}{skipdef}
\bicsintabular{ŘŽŮŘ§Ř¨ŰŘŻŮ}{sl}
\bicsintabular{ŮžŰŘ´âŮŘąŘśâŘŽŮ}{sldefault}
\bicsintabular{Ř´ÚŠŮâŘŽŮŘ§Ř¨ŰŘŻŮ}{slshape}
\bicsintabular{ŘąŘ§Ř­ŘŞâÚŰŮ}{sloppy}
\bicsintabular{Ř´ŮŘ§ŰŮâŘŽŮ}{slshape}
\bicsintabular{ÚŠŮÚÚŠ}{small}
\bicsintabular{ŮžŘąŘ´âÚŠŮŘŞŘ§Ů}{smallskip}
\bicsintabular{ŮŮŘŻŘ§ŘąŮžŘąŘ´âÚŠŮŘŞŘ§Ů}{smallskipamount}
\bicsintabular{ÚŠŮŘ¨}{smash}
\bicsintabular{ŮŘ¨ŘŽŮŘŻ}{smile}
\bicsintabular{ÚŠŘŻŮÚŠŘ§ŮâŮŮŘ§ŮŘˇŮŘą}{snglfntlocatecode}
\bicsintabular{ŮŘśŘ§}{space}
\bicsintabular{ŘśŘąŰŘ¨âŮŘ§ŘľŮŮ}{spacefactor}
\bicsintabular{ŮŘ§ŘľŮŮâÚŠŮŮŘ§ŘŞ}{spaceskip}
\bicsintabular{ŮžŰÚŠ}{spadesuit}
\bicsintabular{Ř§ŘŻŘşŘ§Ů}{span}
\bicsintabular{ŮŰÚŮ}{special}
\bicsintabular{Ř­ŘŻŘ§ÚŠŘŤŘąŘšŮŮâŘłŘŞŮŮ}{splitmaxdepth}
\bicsintabular{ŮŘ§ŘľŮŮâŘ¨Ř§ŮŘ§ŰâŘłŘŞŮŮ}{splittopskip}
\bicsintabular{ŘłŘŞŘ§ŘąŮ}{star}
\bicsintabular{ÚŻŘ§ŮâŘ´ŮŘ§ŘąŮŘŻŮ}{stepcounter}
\bicsintabular{ÚŠŘ´Ű}{stretch}
\bicsintabular{ŘąŘ´ŘŞŮ}{string}
\bicsintabular{Ř´ŮŘš}{strut}
\bicsintabular{ÚŠŘ§ŘŻŘąŘ´ŮŘš}{strutbox}
\bicsintabular{Ř˛ŰŘąŘ¨ŮŘŻ}{subitem}
\bicsintabular{Ř˛ŰŘąŮžŘ§ŘąŘ§ÚŻŘąŘ§Ů}{subparagraph}
\bicsintabular{Ř˛ŰŘąŮŘłŮŘŞ}{subsection}
\bicsintabular{Ř˛ŰŘąŮžŘ´ŘŞŮ}{substack}
\bicsintabular{Ř˛ŰŘąŘ˛ŰŘąŘ¨ŮŘŻ}{subsubitem}
\bicsintabular{Ř˛ŰŘąŘ˛ŰŘąŮŘłŮŘŞ}{subsubsection}
\bicsintabular{Ř˛ŰŘąŮŘŹŮŮŘšŮ}{subset}
\bicsintabular{Ř˛ŰŘąŮŘŹŮŮŘšŮâŮŘł}{subseteq}
\bicsintabular{ŮŮŘŞŮŘ§ŰâŘľŮŘ­Ů}{supereject}
\bicsintabular{Ř­Ř°ŮâŮÚŠŘ§ŮâŘ´ŮŘ§ŮŘą}{suppressfloats}
\bicsintabular{@ŮŮŮŘŞâŘ˘}{@tempa}
\bicsintabular{@ŮŮŮŘŞâŘ¨}{@tempb}
\bicsintabular{@ŮŮŮŘŞâŮž}{@tempc}
\bicsintabular{@ŮŮŮŘŞâŘŞ}{@tempd}
\bicsintabular{@ŮŮŮŘŞâŘŤ}{@tempe}
\bicsintabular{@ÚŠŘ§ŘŻŘąŮŘŞâŘ˘}{@tempboxa}
\bicsintabular{@Ř´ŮŘ§ŮŘŞâŘ˘}{@tempcnta}
\bicsintabular{@Ř´ŮŘ§ŮŘŞâŘ¨}{@tempcntb}
\bicsintabular{@Ř¨ŘšŘŻŮŘŞâŘ˘}{@tempdima}
\bicsintabular{@Ř¨ŘšŘŻŮŘŞâŘ¨}{@tempdimb}
\bicsintabular{@Ř¨ŘšŘŻŮŘŞâŮž}{@tempdimc}
\bicsintabular{@ŮŘ§ŮŘŞâŘ˘}{@tempskipa}
\bicsintabular{@ŮŘ§ŮŘŞâŘ¨}{@tempskipb}
\bicsintabular{@ŘłŮŘ§ŮŘŞâŘ˘ŮŘ§ŘŻŘąŘłŘŞ}{@tempswafalse}
\bicsintabular{@ŘłŮŘ§ŮŘŞâŘ˘ŘŻŘąŘłŘŞ}{@tempswatrue}
\bicsintabular{@ŘŹŘ˛ŮŘŞâŘ˘}{@temptokena}
\bicsintabular{Ř§ŮÚŻâŘ˛ŰŘąŮŮŰŘł}{@thefnmark}
\bicsintabular{@ŘłŮŮŰâŘ§Ř˛ŘłŮ}{@thirdofthree}
\bicsintabular{ŮŘ§ŘľŮŮâŘŹŘ§ÚŻŘ°Ř§ŘąŰ}{tabbingsep}
\bicsintabular{ŮŘ§ŘľŮŮâŘ¨ŰŮâŘłŘŞŮŮŮŘ§}{tabcolsep}
\bicsintabular{ŮŮŘąŘłŘŞâŮŘˇŘ§ŮŘ¨}{tableofcontents}
\bicsintabular{ŘšŮŮŘ§ŮâŘŹŘŻŮŮ}{tablename}
\bicsintabular{ŮŘ§ŘľŮŮâŘłŘŞŮŮŮŘ§}{tabskip}
\bicsintabular{ŘŞŮâŘłŘˇŘąŘŹŘŻŮŮ}{tabularnewline}
\bicsintabular{Ř§ŘŞŰÚŠŘŞ}{tag}
\bicsintabular{ŘŞŮŮŮ}{telephone}
\bicsintabular{ŘŞÚŠ}{TeX}
\bicsintabular{ŮŘŞŮ}{text}
\bicsintabular{ÚŻŮŮŮŮâŮŘŞŮŰ}{textbullet}
\bicsintabular{ŮŮŮâŮŘŞŮ}{textfont}
\bicsintabular{Ř§ŮâŘŻŘ´âŮŘŞŮŰ}{textemdash}
\bicsintabular{Ř§ŮâŘŻŘ´âŮŘŞŮŰ}{textendash}
\bicsintabular{ŘŞŘšŘŹŘ¨âŮŘ§ŘąŮŮŮâŮŘŞŮŰ}{textexclamdown}
\bicsintabular{ŮŮŘˇŮâŮŘłŘˇâŮŘŞŮŰ}{textperiodcentered}
\bicsintabular{ŘłŮŘ§ŮâŮŘ§ŘąŮŮŮâŮŘŞŮŰ}{textquestiondown}
\bicsintabular{ŮŮŮâÚŮžâŮŘŞŮŰâŘŻŮŮŘ§}{textquotedblleft}
\bicsintabular{ŮŮŮâŘąŘ§ŘłŘŞâŮŘŞŮŰâŘŻŮŮŘ§}{textquotedblright}
\bicsintabular{ŮŮŮâŮŘŞŮŰâÚŮž}{textquoteleft}
\bicsintabular{ŮŮŮâŮŘŞŮŰâŘąŘ§ŘłŘŞ}{textquoteright}
\bicsintabular{ŮŘśŘ§ŰâŮŮŘ§ŰŘ§ŮâŮŘŞŮŰâ}{textvisiblespace}
\bicsintabular{Ř´ÚŠŘ§ŮŘŞâŮžŘ´ŘŞâŮŘŞŮŰ}{textbackslash}
\bicsintabular{ŮŰŮŮâŮŘŞŮŰ}{textbar}
\bicsintabular{Ř¨Ř˛ŘąÚŻâŘŞŘąâŮŘŞŮŰ}{textgreater}
\bicsintabular{ÚŠŮŘŞŘąâŮŘŞŮŰ}{textless}
\bicsintabular{ŮŘŞŮâŘłŰŘ§Ů}{textbf}
\bicsintabular{ŮŘŻŮŘąâŮŘŞŮŰ}{textcircled}
\bicsintabular{ŘąŮÚŻâŮŘŞŮ}{textcolor}
\bicsintabular{ŮŘ´Ř§ŮâÚŠŮŮŮâŮŘąÚŠŘ¨âŮŘŞŮ}{textcompwordmark}
\bicsintabular{ŮŘ§ŘľŮŮâŮŘŞŮâŮŘ´ŮŘ§ŮŘą}{textfloatsep}
\bicsintabular{ŮŘłŘ¨ŘŞâŮŘŞŮ}{textfraction}
\bicsintabular{Ř¨ŮŮŘŻŘ§ŰâŮŘŞŮ}{textheight}
\bicsintabular{ŮŘŞŮâŘŞŮŘąŮŘŞŮ}{textindent}
\bicsintabular{ŮŘŞŮâŘ§ŰŘŞŘ§ŮŰÚŠ}{textit}
\bicsintabular{ŮŘŞŮâŮŘ§Ř˛ÚŠ}{textmd}
\bicsintabular{ŮŘŞŮâŮŘąŮŘ§Ů}{textnormal}
\bicsintabular{ŘŤŘ¨ŘŞŰâŮŘŞŮŰ}{textregistered}
\bicsintabular{ŮŘŞŮâŘąŮŮŮ}{textrm}
\bicsintabular{ŮŘŞŮâŘŞŮŘ§ŮâŘ¨Ř˛ŘąÚŻ}{textsc}
\bicsintabular{ŮŘŞŮâŘłŮâŘłŘąŰŮ}{textsf}
\bicsintabular{ŮŘŞŮâŘŽŮŘ§Ř¨ŰŘŻŮ}{textsl}
\bicsintabular{ŘłŘ¨ÚŠâŮŘŞŮŰ}{textstyle}
\bicsintabular{Ř¨Ř§ŮŘ§ŮŮŰŘłâŮŘŞŮŰ}{textsuperscript}
\bicsintabular{ŘšŮŘ§ŮŘŞâŘŞŘŹŘ§ŘąŰâŮŘŞŮŰ}{texttrademark}
\bicsintabular{ŮŘŞŮâŘŞŘ§ŰŮž}{texttt}
\bicsintabular{ŮŘŞŮâŘ§ŰŘłŘŞŘ§ŘŻŮ}{textup}
\bicsintabular{ŮžŮŮŘ§ŰâŮŘŞŮ}{textwidth}
\bicsintabular{Ř˛ŰŘąâŮŮŰŘłâŘšŮŮŘ§Ů}{thanks}
\bicsintabular{ŮŘ­ŘŞŮŘ§Ű}{the}
\bicsintabular{Ř§ŰŮâŘ˛ŰŘąŮŮŰŘł}{thempfn}
\bicsintabular{ŘŽŘˇâŮŘ§ŘśŘŽŰŮ}{thicklines}
\bicsintabular{ŮŘ§ŘľŮŮâŘ˛ŰŘ§ŘŻŘąŰŘ§ŘśŰ}{thickmuskip}
\bicsintabular{ŮŘ§ŘľŮŮâÚŠŮâŘąŰŘ§ŘśŰ}{thinmuskip}
\bicsintabular{ŮŘśŘ§ŘśŘŽŰŮ}{thickspace}
\bicsintabular{ŘŽŘˇâŮŘ§ŮŘ§Ř˛ÚŠ}{thinlines}
\bicsintabular{ŘŻŮŘąŰâÚŠŮÚÚŠ}{thinspace}
\bicsintabular{Ř§ŰŮâŘľŮŘ­ŮâŘŞŘŹŮŮŰ}{thisfancypage}
\bicsintabular{ŘłŘ¨ÚŠâŘ§ŰŮâŘľŮŘ­Ů}{thispagestyle}
\bicsintabular{ŘłŮ@}{thr@@}
\bicsintabular{ŮŘŻ}{tilde}
\bicsintabular{Ř¸ŘąŰŮ}{tiny}
\bicsintabular{Ř˛ŮŘ§Ů}{time}
\bicsintabular{ŘśŘąŘ¨âŘŻŘą}{times}
\bicsintabular{ŘšŮŮŘ§Ů}{title}
\bicsintabular{Ř¨Ů}{to}
\bicsintabular{Ř§ŮŘąŮŘ˛}{today}
\bicsintabular{ŘŹŘ˛ŘĄ}{toks}
\bicsintabular{ŘŞŘšŘąŰŮâŘŹŘ˛ŘĄ}{toksdef}
\bicsintabular{Ř­ŘŻŘ¨ŘŻŮŮŘ§ŰŰ}{tolerance}
\bicsintabular{Ř¨Ř§ŮŘ§}{top}
\bicsintabular{ŘŽŘˇâŘ¨Ř§ŮŘ§ŰâŘ´ŮŘ§ŮŘą}{topfigrule}
\bicsintabular{ŮŘłŘ¨ŘŞâŘ¨Ř§ŮŘ§}{topfraction}
\bicsintabular{Ř­Ř§Ř´ŰŮâŘ¨Ř§ŮŘ§}{topmargin}
\bicsintabular{ŘšŮŘ§ŮŘŞâŘ¨Ř§ŮŘ§}{topmark}
\bicsintabular{ÚŠŘ§ŘŻŘąŘŞŘ§Ř¨Ř§ŮŘ§}{toppageskip}
\bicsintabular{ŮŘ§ŘľŮŮâŘ¨Ř§ŮŘ§ŰâŮŰŘłŘŞ}{topsep}
\bicsintabular{ŮŘ§ŘľŮŮâŘ¨Ř§ŮŘ§}{topskip}
\bicsintabular{Ř¨ŮŮŘŻŘ§ŰâÚŠŮ}{totalheight}
\bicsintabular{ŘąŘŻÚŻŰŘąŰâÚŠŮ}{tracingall}
\bicsintabular{ŘąŘŻÚŻŰŘąŰâŮŘąŘ§ŮŰŮ}{tracingcommands}
\bicsintabular{ŘąŘŻÚŻŰŘąŰâŘ­ŘąŮŮ}{tracinglostchars}
\bicsintabular{ŘąŘŻÚŻŰŘąŰâŮŘ§ÚŠŘąŮŮŘ§}{tracingmacros}
\bicsintabular{ŘąŘŻÚŻŰŘąŰâŮŮŘ§ŰŘ´Ű}{tracingonline}
\bicsintabular{ŘąŘŻÚŻŰŘąŰâŘľŮŘ­ŮâŘ¨ŮŘŻŰ}{tracingoutput}
\bicsintabular{ŘąŘŻÚŻŰŘąŰâŘľŮŘ­Ř§ŘŞ}{tracingpages}
\bicsintabular{ŘąŘŻÚŻŰŘąŰâŘ¨ŮŘŻŮŘ§}{tracingparagraphs}
\bicsintabular{ŘąŘŻÚŻŰŘąŰâŘ¨Ř§Ř˛ÚŻŘąŘŻŘ§ŮŰ}{tracingrestores}
\bicsintabular{ŘąŘŻÚŻŰŘąŰâŘ˘ŮŘ§ŘąŮŘ§}{tracingstats}
\bicsintabular{ŮŘŤŮŘŤ}{triangle}
\bicsintabular{ŮžŰŘ´âŮŘąŘśâŘŞŘ§ŰŮž}{ttdefault}
\bicsintabular{ŮŘ§ŮŰŮâŘŞŘ§ŰŮž}{ttfamily}
\bicsintabular{ŘŻŮ@}{tw@}
\bicsintabular{ŘŻŮŘłŘŞŮŮ}{twocolumn}
\bicsintabular{ŘŻŘąŮŮŰŘł}{typein}
\bicsintabular{Ř¨ŘąŮŮŰŘł}{typeout}
\bicsintabular{ÚŠŘŻŘ¨Ř˛ŘąÚŻ}{uccode}
\bicsintabular{ŘŞŰŘąŮâŘ¨ŮŘŻŰâŘ¨Ř˛ŘąÚŻ}{uchyph}
\bicsintabular{ŘŞŘšŘąŰŮâŮŘ´ŘŻŮ}{undefined}
\bicsintabular{Ř˛ŰŘąŘŽŘˇ}{underline}
\bicsintabular{Ř¨ŰâÚŠŘ§ŘŻŘąŘ§}{unhbox}
\bicsintabular{Ř¨ŰâÚŠŮžŰâŘ§}{unhcopy}
\bicsintabular{ŮŘ§Ř­ŘŻŘˇŮŮ}{unitlength}
\bicsintabular{Ř¨ŘąÚŻŘ´ŘŞâŘŻŮŘąŰ}{unkern}
\bicsintabular{Ř¨ŘąÚŻŘ´ŘŞâŘŹŘąŰŮŮ}{unpenalty}
\bicsintabular{Ř¨ŘąÚŻŘ´ŘŞâŮŘ§ŘľŮŮ}{unskip}
\bicsintabular{Ř¨ŰâÚŠŘ§ŘŻŘąŮ}{unvbox}
\bicsintabular{Ř¨ŰâÚŠŮžŰâŮ}{unvcopy}
\bicsintabular{ŮžŰŘ´âŮŘąŘśâŘ§ŰŘłŘŞŘ§ŘŻŮ}{updefault}
\bicsintabular{Ř´ÚŠŮâŘ§ŰŘłŘŞŘ§ŘŻŮ}{upshape}
\bicsintabular{Ř§Ř˛ÚŠŘ§ŘŻŘą}{usebox}
\bicsintabular{Ř¨Ř§Ř´ŮŘ§ŘąŘ´ÚŻŘą}{usecounter}
\bicsintabular{ÚŻŘ˛ŰŮŘ´âŮŮŮ}{usefont}
\bicsintabular{ŘłŘ¨ÚŠâŮŘ§Ř˛Ů}{usepackage}
\bicsintabular{@ŮŘśŘ§ŮŘ§ŰâŮŘšŘ§Ů}{@vobeyspaces}
\bicsintabular{@ŘŞŮŰ}{@void}
\bicsintabular{ŘŞŮŘ¸ŰŮâŮ}{vadjust}
\bicsintabular{ŘąŘŻŰŮâŮ}{valign}
\bicsintabular{ŮŘ­ŘŞŮŘ§ŰâŘ´ŮŘ§ŘąŮŘŻŮ}{value}
\bicsintabular{Ř¨ŘŻŮŮŘ§ŰŰâŮ}{vbadness}
\bicsintabular{ÚŠŘ§ŘŻŘąŮ}{vbox}
\bicsintabular{ÚŠŘ§ŘŻŘąŮŘłŘˇ}{vcenter}
\bicsintabular{ŮŮŘ§ŮŘˇŮŘą}{verb}
\bicsintabular{ŮžŘąŮ}{vfil}
\bicsintabular{ŮžŘąŘąŮ}{vfill}
\bicsintabular{ŘąŮŘšâŮžŘąŮ}{vfilneg}
\bicsintabular{ŮžŘąŘ˛ŘšŮŮŘŻŰ}{vfuzz}
\bicsintabular{ŮŮŘ§ŰŘ§Ů}{visible}
\bicsintabular{ŘŽŘˇâŘšŮŮŘŻ}{vline}
\bicsintabular{Ř­Ř§Ř´ŰŮâŮ}{voffset}
\bicsintabular{ÚŠ@ŘŻŘąŘŞŮŰ}{voidb@x}
\bicsintabular{Ř§ŘąŘŹŘ§ŘšâŘľŮŘ­ŮâŘš}{vpageref}
\bicsintabular{ŮŘ§ŘľŮŮâŮÚŻŘąŘŻ}{vrboxsep}
\bicsintabular{Ř§ŘąŘŹŘ§ŘšâŘš}{vref}
\bicsintabular{ŘŽŘˇâŮ}{vrule}
\bicsintabular{ŘˇŮŮâŘľŮŘ­Ů}{vsize}
\bicsintabular{ŮŘ§ŘľŮŮâŮ}{vskip}
\bicsintabular{ŮŘśŘ§ŰâŮ}{vspace}
\bicsintabular{Ř´ÚŠŘłŘŞâŮ}{vsplit}
\bicsintabular{ŮŘąŘŻŮŮ}{vss}
\bicsintabular{ÚŠŘ§ŘŻŘąÚŻŮŘŻ}{vtop}
\bicsintabular{ŘšŘąŘś}{wd}
\bicsintabular{ŮŘ§ŘŻŘ§ŮâŘ¨ÚŠŮ}{whiledo}
\bicsintabular{ÚŠŮŘ§ŮâŮžŮŮ}{widehat}
\bicsintabular{ŮŘŻŮžŮŮ}{widetilde}
\bicsintabular{ŘŹŘąŰŮŮâŘŞŮâŘ¨ŮŘŻ}{widowpenalty}
\bicsintabular{ŮžŮŮŘ§}{width}
\bicsintabular{ŘŻŘąÚŠŘ§ŘąŮŘ§ŮŮ}{wlog}
\bicsintabular{Ř¨ŮŮŰŘł}{write}
\bicsintabular{@ŮŘśŘ§ŰâŮŘ§ŘŞŰŮ}{@xobeysp}
\bicsintabular{ŘłŰ@ŘŻŮ}{@xxxii}
\bicsintabular{ŘŞŘąÚŻŘš}{xdef}
\bicsintabular{ŮŘ´Ř§ŮÚŻŘąÚŻŘłŘŞŘąŘ´Ű}{xleaders}
\bicsintabular{ŮŘ§ŘľŮŮâŘ§ŘśŘ§ŮŰâÚŠŮŮŘ§ŘŞ}{xspaceskip}
\bicsintabular{ŘłŘ§Ů}{year}
\bicsintabular{@ŮŘą}{z@}
\bicsintabular{@ŮŘąŮŘ§ŘľŮŮ}{z@skip}
\end{supertabular}
\end{center}

\bigskip
\begin{center}
\tablecaption{The Equivalent \XePersian\ Commands\label{xcs}}
\tablehead
   {\bfseries Command in \XePersian &\bfseries  Equivalent Persian Command\\ \hline}
\tabletail
   {\hline \multicolumn{2}{r}{\emph{Continued on next page}}\\}
\tablelasttail{\hline}
\begin{supertabular}{lr}
\bicsintabular{ŘŽŘˇâŘ˛ŰŘąŮŮŰŘłâŘŽŮŘŻÚŠŘ§Řą}{autofootnoterule}
\bicsintabular{Ř§ŘšŘŻŘ§ŘŻŮŘąŮŮŮŮŘ§ŘŽŮŘŻÚŠŘ§Řą}{AutoMathDigits}
\bicsintabular{Ř§ŘšŘŻŘ§ŘŻŮŘąŮŮŮŮŘ§ŮŘ§ŘŞŰŮ}{DefaultMathDigits}
\bicsintabular{ŘŞŘšŘąŰŮâŮŮŮâŮŘ§ŘŞŰŮ}{deflatinfont}
\bicsintabular{ŘŞŘšŘąŰŮâŮŮŮâŮžŘ§ŘąŘłŰ}{defpersianfont}
\bicsintabular{ÚŠŘ§ŘŻŘąŘ§ÚŮž}{hboxL}
\bicsintabular{ÚŠŘ§ŘŻŘąŘ§ŘłŘŞ}{hboxR}
\bicsintabular{ŮŘšŘ§ŘŻŮ@ÚŠŮŰŘŻ}{keyval@eq@alias@key}
\bicsintabular{ŮŘąŘŹŘšâŮŘ§ŘŞŰŮ}{Latincite}
\bicsintabular{ŮŮŮâŮŘ§ŘŞŰŮ}{latinfont}
\bicsintabular{Ř§ŮŘąŮŘ˛ŮŘ§ŘŞŰŮ}{latintoday}
\bicsintabular{ŘŽŘˇâŘ˛ŰŘąŮŮŰŘłâÚŮž}{leftfootnoterule}
\bicsintabular{ŮŘŞŮâŮŘ§ŘŞŰŮ}{lr}
\bicsintabular{ÚŮžâŘ¨ŘąŘ§ŘłŘŞ}{LRE}
\bicsintabular{ŘŻŮŘłŘŞŮŮŰâÚŮž}{LTRdblcol}
\bicsintabular{ŮžŘ§ŮŮŰŘł}{LTRfootnote}
\bicsintabular{ŮŘŞŮâŮžŘ§ŮŮŰŘł}{LTRfootnotetext}
\bicsintabular{ŮžŘ§ŮŮŰŘłâŘšŮŮŘ§Ů}{LTRthanks}
\bicsintabular{ŘąŮŘ˛ŮžŘ§ŘąŘłŰ}{persianday}
\bicsintabular{ŮŮŮâŮžŘ§ŘąŘłŰ}{persianfont}
\bicsintabular{Ř§ŘšŘŻŘ§ŘŻŮŘąŮŮŮŮŘ§ŮžŘ§ŘąŘłŰ}{PersianMathDigits}
\bicsintabular{ŮŘ§ŮâŮžŘ§ŘąŘłŰ}{persianmonth}
\bicsintabular{ŘłŘ§ŮâŮžŘ§ŘąŘłŰ}{persianyear}
\bicsintabular{ŘšŮŘ§ŮŘŞâÚŮžâŮŮŮâŮŮŮâŮžŘ§ŘąŘłŰâ}{plq}
\bicsintabular{ŘšŮŘ§ŮŘŞâŘąŘ§ŘłŘŞâŮŮŮâŮŮŮâŮžŘ§ŘąŘłŰ}{prq}
\bicsintabular{ŘŽŘˇâŘ˛ŰŘąŮŮŰŘłâŘąŘ§ŘłŘŞ}{rightfootnoterule}
\bicsintabular{ŮŘŞŮâŮžŘ§ŘąŘłŰ}{rl}
\bicsintabular{ŘąŘ§ŘłŘŞâŘ¨ÚŮž}{RLE}
\bicsintabular{ŘŻŮŘłŘŞŮŮŰâŘąŘ§ŘłŘŞ}{RTLdblcol}
\bicsintabular{ŮžŘ§ŮŮŘ´ŘŞ}{RTLfootnote}
\bicsintabular{ŮŘŞŮâŮžŘ§ŮŮŘ´ŘŞ}{RTLfootnotetext}
\bicsintabular{ŮžŘ§ŮŮŘ´ŘŞâŘšŮŮŘ§Ů}{RTLthanks}
\bicsintabular{@ŘšŮŘ§ŮŘŞâŘ¨ŰŮ}{@SepMark}
\bicsintabular{ŘšŮŘ§ŮŘŞâŘ¨ŰŮ}{SepMark}
\bicsintabular{Ř¨ÚŻŘ°Ř§ŘąŮŘąŘŹŮŘšŘ§ŘŞâŘšŘ§ŘŻŰ}{setdefaultbibitems}
\bicsintabular{Ř¨ÚŻŘ°Ř§ŘąŘŻŘąŘ­Ř§Ř´ŰŮâŘšŘ§ŘŻŰ}{setdefaultmarginpar}
\bicsintabular{ÚŻŘ˛ŰŮŘ´âŮŮŮâŘ§ŘšŘŻŘ§ŘŻŮŘąŮŮŮŮŘ§}{setmathdigitfont}
\bicsintabular{Ř¨ÚŻŘ°Ř§ŘąŘ˛ŰŘąŮŮŰŘłâÚŮž}{setfootnoteLR}
\bicsintabular{Ř¨ÚŻŘ°Ř§ŘąŘ˛ŰŘąŮŮŰŘłâŘąŘ§ŘłŘŞ}{setfootnoteRL}
\bicsintabular{ÚŻŘ˛ŰŮŘ´âŮŮŮâŮŘ§ŘŞŰŮâŮŘŞŮ}{setlatintextfont}
\bicsintabular{Ř¨ÚŻŘ°Ř§ŘąŮŘŞŮâÚŮž}{setLTR}
\bicsintabular{Ř¨ÚŻŘ°Ř§ŘąŮŘąŘŹŮŘšŘ§ŘŞâÚŮž}{setLTRbibitems}
\bicsintabular{Ř¨ÚŻŘ°Ř§ŘąŘŻŘąŘ­Ř§Ř´ŰŮâÚŮž}{setLTRmarginpar}
\bicsintabular{Ř¨ÚŻŘ°Ř§ŘąŮŘŞŮâŘąŘ§ŘłŘŞ}{setRTL}
\bicsintabular{Ř¨ÚŻŘ°Ř§ŘąŮŘąŘŹŮŘšŘ§ŘŞâŘąŘ§ŘłŘŞ}{setRTLbibitems}
\bicsintabular{Ř¨ÚŻŘ°Ř§ŘąŘŻŘąŘ­Ř§Ř´ŰŮâŘąŘ§ŘłŘŞ}{setRTLmarginpar}
\bicsintabular{ÚŻŘ˛ŰŮŘ´âŮŮŮâŮŘŞŮ}{settextfont}
\bicsintabular{ŘŽŘˇâŘ˛ŰŘąŮŮŰŘłâŮžŮŮŘ§ŰâŮŘŞŮ}{textwidthfootnoterule}
\bicsintabular{ŮŮŘąŘłŘŞâŮŘˇŘ§ŮŘ¨âŘŻŮŘłŘŞŮŮŰ}{twocolumnstableofcontents}
\bicsintabular{ŮÚŻŘ°Ř§ŘąŘ˛ŰŘąŮŮŰŘłâŘąŘ§ŘłŘŞ}{unsetfootnoteRL}
\bicsintabular{ŮÚŻŘ°Ř§ŘąŮŘŞŮâÚŮž}{unsetLTR}
\bicsintabular{ŮÚŻŘ°Ř§ŘąŮŘŞŮâŘąŘ§ŘłŘŞ}{unsetRTL}
\bicsintabular{ÚŠŘ§ŘŻŘąŮŘ§Ř˛ÚŮž}{vboxL}
\bicsintabular{ÚŠŘ§ŘŻŘąŮŘ§Ř˛ŘąŘ§ŘłŘŞ}{vboxR}
\bicsintabular{Ř˛ŰâŮŘ§ŘŞÚŠ}{XeLaTeX}
\bicsintabular{Ř˛ŰâŮžŘąŘ´ŰŮ}{XePersian}
\bicsintabular{ÚŻŮŮŮâŘ˛ŰâŮžŘąŘ´ŰŮ}{xepersianversion}
\bicsintabular{ŘŞŘ§ŘąŰŘŽâÚŻŮŮŮâŘ˛ŰâŮžŘąŘ´ŰŮ}{xepersiandate}
\bicsintabular{Ř˛ŰâŘŞÚŠ}{XeTeX}
\end{supertabular}
\end{center}

\bigskip
\begin{center}
\tablecaption{The Equivalent \LaTeX\ Environments\label{lenv}}
\tablehead
   {\bfseries Environment in  \LaTeX\ &\bfseries  Equivalent Persian Environment\\ \hline}
\tabletail
   {\hline \multicolumn{2}{r}{\emph{Continued on next page}}\\}
\tablelasttail{\hline}
\begin{supertabular}{lr}
\bienvintabular{ÚÚŠŰŘŻŮ}{abstract}
\bienvintabular{ŮžŰŮŘłŘŞ}{appendix}
\bienvintabular{Ř˘ŘąŘ§ŰŮ}{array}
\bienvintabular{ŮŘłŘˇâÚŰŮ}{center}
\bienvintabular{ŘŞŮŘśŰŘ­}{description}
\bienvintabular{ŘąŰŘ§ŘśŰâŮŮŘ§ŰŘ´Ű}{displaymath}
\bienvintabular{ŮŮŘ´ŘŞŘ§Řą}{document}
\bienvintabular{Ř´ŮŘ§ŘąŘ´}{enumerate}
\bienvintabular{Ř´ÚŠŮ}{figure}
\bienvintabular{Ř´ÚŠŮ*}{figure*}
\bienvintabular{ŮŘ­ŘŞŮŘ§ŰâŮžŘąŮŮŘŻŮ}{filecontents}
\bienvintabular{ŮŘ­ŘŞŮŘ§ŰâŮžŘąŮŮŘŻŮ*}{filecontents*}
\bienvintabular{ÚŮžâÚŰŮ}{flushleft}
\bienvintabular{ŘąŘ§ŘłŘŞâÚŰŮ}{flushright}
\bienvintabular{ŮŮŘąŘ§ŘŞ}{itemize}
\bienvintabular{ŮŘ§ŮŮ}{letter}
\bienvintabular{ŮŰŘłŘŞ}{list}
\bienvintabular{ŘŹŘŻŮŮâŘŻŘąŘ§Ř˛}{longtable}
\bienvintabular{ÚŠŘ§ŘŻŘąŘąÚ}{lrbox}
\bienvintabular{ŘąŰŘ§ŘśŰ}{math}
\bienvintabular{ŮŘ§ŘŞŘąŰŘł}{matrix}
\bienvintabular{ŘľŮŘ­ŮâÚŠŮÚÚŠ}{minipage}
\bienvintabular{ÚŮŘŻŘłŘŞŮŮŰâŮŘ§}{multicols}
\bienvintabular{ÚŮŘŻŘŽŘˇŰ}{multline}
\bienvintabular{ŰŘ§ŘŻŘŻŘ§Ř´ŘŞ}{note}
\bienvintabular{Ř§ŮŘ¨Ř§Ř´ŘŞŮ}{overlay}
\bienvintabular{ŘŞŘľŮŰŘą}{picture}
\bienvintabular{Ř§ŮŘŞŘ¨Ř§Řł}{quotation}
\bienvintabular{ŮŮŮ}{quote}
\bienvintabular{Ř§ŘłŮŘ§ŰŘŻ}{slide}
\bienvintabular{ŮžŘ§ŘąŮŘ§ŮŘąŘŞŘ¨}{sloppypar}
\bienvintabular{Ř´ÚŠŘ§ŮŘŞŮ}{split}
\bienvintabular{Ř˛ŰŘąŘ˘ŘąŘ§ŰŮ}{subarray}
\bienvintabular{ŘŹŘ§ÚŻŘ°Ř§ŘąŰ}{tabbing}
\bienvintabular{ŮŮŘ­}{table}
\bienvintabular{ŮŮŘ­*}{table*}
\bienvintabular{ŘŹŘŻŮŮ}{tabular}
\bienvintabular{ŘŹŘŻŮŮ*}{tabular*}
\bienvintabular{ŮŘąŘ§ŘŹŘš}{thebibliography}
\bienvintabular{ŮŘ­ŘŞŮŘ§ŰâŮŮŘ§ŰŮ}{theindex}
\bienvintabular{ŘľŮŘ­ŮâŘšŮŮŘ§Ů}{titlepage}
\bienvintabular{ŮŰŘłŘŞâŘ¨ŘŻŮŰ}{trivlist}
\bienvintabular{ŮŮŘ§ŮŘˇŮŘąÚŠŮâŮŘłŘŞ}{verbatim}
\bienvintabular{ŮŮŘ§ŮŘˇŮŘąÚŠŮâŮŘłŘŞ*}{verbatim*}
\bienvintabular{Ř´ŘšŘą}{verse}
\end{supertabular}
\end{center}

\bigskip
\begin{center}
\tablecaption{The Equivalent \XePersian\ Environments\label{xenv}}
\tablehead
   {\bfseries Environment in  \XePersian\ &\bfseries  Equivalent Persian Environment\\ \hline}
\tabletail
   {\hline \multicolumn{2}{r}{\emph{Continued on next page}}\\}
\tablelasttail{\hline}
\begin{supertabular}{lr}
\bienvintabular{ŮŘ§ŘŞŰŮ}{latin}
\bienvintabular{ŮŘŞŮâÚŮž}{LTR}
\bienvintabular{ŘŻŘłŘŞŮâŘ¨ŮŘŻŰâÚŮž}{LTRitems}
\bienvintabular{ŮžŘ§ŘąŘłŰ}{persian}
\bienvintabular{ŮŘŞŮâŘąŘ§ŘłŘŞ}{RTL}
\bienvintabular{ŘŻŘłŘŞŮâŘ¨ŮŘŻŰâŘąŘ§ŘłŘŞ}{RTLitems}
\end{supertabular}
\end{center}
\paragraph{Localisation of postion arguments.}
 Some environments like \texttt{tabular} and some commands like \Lcs{parbox} have an argument which specifies the position. Table \autoref{table-pos} shows their localisations.
\begin{center}
\tablecaption{The Equivalent \LaTeX{} position arguments\label{table-pos}}
\tablehead
   {\bfseries Position argument in \LaTeX &\bfseries  Equivalent Persian position argument\\ \hline}
\tabletail
   {\hline \multicolumn{2}{r}{\emph{Continued on next page}}\\}
\tablelasttail{\hline}
\begin{supertabular}{cc}
\texttt{b}&\Penv{Ř˛}\\
\texttt{c}&\Penv{Ů}\\
\texttt{C}&\Penv{Řł}\\
\texttt{h}&\Penv{Ř§}\\
\texttt{H}&\Penv{Ř˘}\\
\texttt{J}&\Penv{ŘŞ}\\
\texttt{l}&\Penv{Ú}\\
\texttt{L}&\Penv{Ů}\\
\texttt{m}&\Penv{Ů}\\
\texttt{p}&\Penv{Ůž}\\
\texttt{p}&\Penv{Řľ}\\
\texttt{r}&\Penv{Řą}\\
\texttt{R}&\Penv{Ř§}\\
\texttt{s}&\Penv{ÚŠ}\\
\texttt{t}&\Penv{Ř¨}\\
\end{supertabular}
\end{center}
\begin{itemize}
\item There are two \texttt{p}s in \autoref{table-pos},  first \texttt{p} and its Persian equivalent \Penv{Ůž} stand for paragraph (used in \texttt{tabular} and similar environments) and the second  \texttt{p} and its Persian equivalent \Penv{Řľ} stand for page (used in \texttt{float}-like environments).
\end{itemize}
\subsubsection{Localizations of the keys and key values of \textsf{graphicx} package}
The equivalent Persian keys and key values of \textsf{graphicx} package is shown in \autoref{table:key} and \autoref{table:keyvalue} respectively.
\begin{center}
\tablecaption{Persian Equivalent keys of \textsf{graphicx} package\label{table:key}}
\tablehead
   {\bfseries Original Key &\bfseries  Equivalent Persian Key\\ \hline}
\tabletail
   {\hline \multicolumn{2}{r}{\emph{Continued on next page}}\\}
\tablelasttail{\hline}
\begin{supertabular}{lr}
\texttt{draft}&\Penv{ŮžŰŘ´âŮŮŰŘł}\\
\texttt{origin}&\Penv{ŮŘ¨ŘŻŘ§}\\
\texttt{clip}&\Penv{Ř¨ŰâŘ§ŘśŘ§ŮŮ}\\
\texttt{keepaspectratio}&\Penv{Ř­ŮŘ¸âŘŞŮŘ§ŘłŘ¨}\\
\texttt{natwidth}&\Penv{ŮžŮŮŘ§ŰâŘˇŘ¨ŰŘšŰ}\\
\texttt{natheight}&\Penv{Ř¨ŮŮŘŻŘ§ŰâŘˇŘ¨ŰŘšŰ}\\
\texttt{bb}&\Penv{ŮŘŽŘŞŘľŘ§ŘŞ}\\
\texttt{viewport}&\Penv{ŮŘ­ŘŻŮŘŻŮâŮŮŘ§ŰŘ´}\\
\texttt{trim}&\Penv{Ř­Ř°ŮâŘ§ŘˇŘąŘ§Ů}\\
\texttt{angle}&\Penv{Ř˛Ř§ŮŰŮ}\\
\texttt{width}&\Penv{ŮžŮŮŘ§}\\
\texttt{height}&\Penv{Ř¨ŮŮŘŻŘ§}\\
\texttt{totalheight}&\Penv{Ř¨ŮŮŘŻŘ§ŰâÚŠŮ}\\
\texttt{scale}&\Penv{ŘśŘąŰŘ¨}\\
\texttt{type}&\Penv{ŮŮŘš}\\
\texttt{ext}&\Penv{ŮžŘłŮŮŘŻ}\\
\texttt{read}&\Penv{ŘŽŮŘ§ŮŘŻŮŰ}\\
\texttt{command}&\Penv{ŮŘąŮŘ§Ů}\\
\texttt{x}&\Penv{ŘˇŮŮ}\\
\texttt{y}&\Penv{ŘšŘąŘś}\\
\texttt{units}&\Penv{ŮŘ§Ř­ŘŻ}\\
\end{supertabular}
\end{center}
\begin{center}
\tablecaption{Persian Equivalent key values of \textsf{graphicx} package\label{table:keyvalue}}
\tablehead
   {\bfseries Original Key value&\bfseries  Equivalent Persian Key value\\ \hline}
\tabletail
   {\hline \multicolumn{2}{r}{\emph{Continued on next page}}\\}
\tablelasttail{\hline}
\begin{supertabular}{cc}
\texttt{b}&\Penv{Ř˛}\\
\texttt{B}&\Penv{ÚŠ}\\
\texttt{false}&\Penv{ŮŘ§ŘŻŘąŘłŘŞ}\\
\texttt{l}&\Penv{Ú}\\
\texttt{r}&\Penv{Řą}\\
\texttt{t}&\Penv{Ř¨}\\
\texttt{true}&\Penv{ŘŻŘąŘłŘŞ}\\
\end{supertabular}
\end{center}
\subsubsection{Localizations of font features and font feature options}
The equivalent Persian font features and font feature options is shown in \autoref{table:ff} and \autoref{table:ffo} respectively.
\begin{center}
\tablecaption{Persian Equivalent font features\label{table:ff}}
\tablehead
   {\bfseries Original font feature &\bfseries  Equivalent Persian font feature\\ \hline}
\tabletail
   {\hline \multicolumn{2}{r}{\emph{Continued on next page}}\\}
\tablelasttail{\hline}
\begin{supertabular}{lr}
\biffintabular{ExternalLocation}{ŮÚŠŘ§ŮâŘŽŘ§ŘąŘŹŰ}
\biffintabular{ExternalLocation}{ŮŘłŰŘą}
\biffintabular{Renderer}{ŘŞŘ­ŮŰŮâŘŻŮŮŘŻŮ}
\biffintabular{BoldFont}{ŮŮŮâŘłŰŘ§Ů}
\biffintabular{Language}{Ř˛Ř¨Ř§Ů}
\biffintabular{Script}{ŘŽŘˇ}
\biffintabular{UprightFont}{ŮŮŮâŘšŮŮŘŻŰ}
\biffintabular{ItalicFont}{ŮŮŮâŘ§ŰŘŞŘ§ŮŰÚŠ}
\biffintabular{BoldItalicFont}{ŮŮŮâŘ§ŰŘŞŘ§ŮŰÚŠâŘłŰŘ§Ů}
\biffintabular{SlantedFont}{ŮŮŮâŘŽŮŘ§Ř¨ŰŘŻŮ}
\biffintabular{BoldSlantedFont}{ŮŮŮâŘŽŮŘ§Ř¨ŰŘŻŮâŘłŰŘ§Ů}
\biffintabular{SmallCapsFont}{ŮŮŮâÚŠŮŘ§ŮâÚŠŮÚÚŠ}
\biffintabular{UprightFeatures}{ŮŰÚÚŻŰâŮŘ§ŰâŮŮŮâŘšŮŮŘŻŰ}
\biffintabular{BoldFeatures}{ŮŰÚÚŻŰâŮŘ§ŰâŮŮŮâŘłŰŘ§Ů}
\biffintabular{ItalicFeatures}{ŮŰÚÚŻŰâŮŘ§ŰâŮŮŮâŘ§ŰŘŞŘ§ŮŰÚŠ}
\biffintabular{BoldItalicFeatures}{ŮŰÚÚŻŰâŮŘ§ŰâŮŮŮâŘ§ŰŘŞŘ§ŮŰÚŠâŘłŰŘ§Ů}
\biffintabular{SlantedFeatures}{ŮŰÚÚŻŰâŮŘ§ŰâŮŮŮâŘŽŮŘ§Ř¨ŰŘŻŮ}
\biffintabular{BoldSlantedFeatures}{ŮŰÚÚŻŰâŮŘ§ŰâŮŮŮâŘŽŮŘ§Ř¨ŰŘŻŮâŘłŰŘ§Ů}
\biffintabular{SmallCapsFeatures}{ŮŰÚÚŻŰâŮŘ§ŰâŮŮŮâÚŠŮŘ§ŮâÚŠŮÚÚŠ}
\biffintabular{SizeFeatures}{ŮŰÚÚŻŰâŮŘ§ŰâŘ§ŮŘŻŘ§Ř˛Ů}
\biffintabular{Scale}{ŘśŘąŰŘ¨}
\biffintabular{WordSpace}{ŮŘśŘ§ŰâÚŠŮŮŮ}
\biffintabular{PunctuationSpace}{ŮŘśŘ§ŰâŮŮŘˇŮâÚŻŘ°Ř§ŘąŰ}
\biffintabular{FontAdjustment}{ŘŞŮŘ¸ŰŮâŮŮŮ}
\biffintabular{LetterSpace}{ŮŘśŘ§ŰâŘ­ŘąŮ}
\biffintabular{HyphenChar}{ŮŮŰŘłŮâŘŞŰŘąŮ}
\biffintabular{Color}{ŘąŮÚŻ}
\biffintabular{Opacity}{ÚŠŘŻŘąŰ}
\biffintabular{Mapping}{ŮÚŻŘ§Ř´ŘŞ}
\biffintabular{Weight}{ŘłŮÚŻŰŮŰ}
\biffintabular{Width}{ŮžŮŮŘ§}
\biffintabular{OpticalSize}{Ř§ŮŘŻŘ§Ř˛ŮâÚŘ´ŮŰ}
\biffintabular{FakeSlant}{ŘŽŮŘ§Ř¨ŰŘŻŮâŘŞŮŮŘ¨Ű}
\biffintabular{FakeStretch}{ÚŠŘ´Ř´âŘŞŮŮŘ¨Ű}
\biffintabular{FakeBold}{ŘłŰŘ§ŮâŘŞŮŮŘ¨Ű}
\biffintabular{AutoFakeSlant}{ŘŽŮŘ§Ř¨ŰŘŻŮâŘŞŮŮŘ¨ŰâŘŽŮŘŻÚŠŘ§Řą}
\biffintabular{AutoFakeBold}{ŘłŰŘ§ŮâŘŞŮŮŘ¨ŰâŘŽŮŘŻÚŠŘ§Řą}
\biffintabular{Ligatures}{ŘŻŮŰŘ§ÚŮŘŻŘ­ŘąŮâŮŘŞŘľŮâŘ¨ŮâŮŮ}
\biffintabular{Alternate}{ŮŘŞŮŘ§ŮŘ¨}
\biffintabular{Variant}{ÚŻŮŮŘ§ÚŻŮŮ}
\biffintabular{Variant}{ŮŘŹŮŮŘšŮâŘłŘ¨ÚŠŰ}
\biffintabular{CharacterVariant}{ÚŻŮŮŘ§ÚŻŮŮŰâŮŮŰŘłŮ}
\biffintabular{Style}{ŘłŘ¨ÚŠ}
\biffintabular{Annotation}{ŰŘ§ŘŻŘŻŘ§Ř´ŘŞ}
\biffintabular{RawFeature}{ŮŰÚÚŻŰâŮŘ§ŰâÚŠŘ§Ů}
\biffintabular{CharacterWidth}{ŮžŮŮŘ§ŰâŮŮŰŘłŮ}
\biffintabular{Numbers}{Ř§ŘąŮŘ§Ů}
\biffintabular{Contextuals}{ŮŘŞŮŰ}
\biffintabular{Diacritics}{ŘŞŮÚŠŰÚŠâÚŠŮŮŘŻŮâŮŘ§}
\biffintabular{Letters}{Ř­ŘąŮŮ}
\biffintabular{Kerning}{ŘŻŮŘąŰ}
\biffintabular{VerticalPosition}{ŮŮŮŘšŰŘŞâŘšŮŮŘŻŰ}
\biffintabular{Fractions}{ÚŠŘłŘą}
\end{supertabular}
\end{center}
\begin{center}
\tablecaption{Persian Equivalent font feature options\label{table:ffo}}
\tablehead
   {\bfseries font feature&\bfseries font feature option &\bfseries  Persian font feature option\\ \hline}
\tabletail
   {\hline \multicolumn{3}{r}{\emph{Continued on next page}}\\}
\tablelasttail{\hline}
\begin{supertabular}{llr}
\biffointabular{Language}{Default}{ŮžŰŘ´âŮŘąŘś}
\biffointabular{Language}{Parsi}{ŮžŘ§ŘąŘłŰ}
\biffointabular{Script}{Parsi}{ŮžŘ§ŘąŘłŰ}
\biffointabular{Script}{Latin}{ŮŘ§ŘŞŰŮ}
\biffointabular{Style}{MathScript}{Ř§ŘłÚŠŘąŰŮžŘŞâŘąŰŘ§ŘśŰ}
\biffointabular{Style}{MathScriptScript}{Ř§ŘłÚŠŘąŰŮžŘŞâŘ§ŘłÚŠŘąŰŮžŘŞâŘąŰŘ§ŘśŰ}
\end{supertabular}
\end{center}
\subsection{A Sample Input \TeX\ File}
\begin{lstlisting}
\documentclass{article}
\usepackage{xepersian}
\settextfont{XB Niloofar}
\title{*\parsitext{ŰÚŠ ŘłŮŘŻ ŮŮŮŮŮ}*}
\author{*\parsitext{ŮŘ§Ů ŮŮŰŘłŮŘŻŮ}*}
\begin{document}
\maketitle
\tableofcontents
\part{*\parsitext{ŘšŮŮŘ§Ů Ř¨ŘŽŘ´}*}
...
\section{*\parsitext{ŘšŮŮŘ§Ů ŮŘłŮŘŞ}*}
...
\subsection{*\parsitext{ŘšŮŮŘ§Ů Ř˛ŰŘąŮŘłŮŘŞ}*}
...
\subsubsection{*\parsitext{ŘšŮŮŘ§Ů Ř˛ŰŘą Ř˛ŰŘąŮŘłŮŘŞ}*}
...
\end{document}
\end{lstlisting}
\subsection{Font Commands}
\subsubsection{Basic Font Commands}
\begin{BDef}
\Lcs{settextfont}\OptArgs\Largb{\Larga{font name}}\\
\Lcs{setlatintextfont}\OptArgs\Largb{\Larga{font name}}\\
\Lcs{setmathdigitfont}\OptArgs\Largb{\Larga{font name}}\\
\Lcs{setmathsfdigitfont}\OptArgs\Largb{\Larga{font name}}\\
\Lcs{setmathttdigitfont}\OptArgs\Largb{\Larga{font name}}
\end{BDef}
\begin{itemize}
\item Options in any font command in this documentation are anything that \textsf{fontspec} package provides as the option of loading fonts, except \texttt{Script} and \texttt{Mapping}.
\item \Lcs{settextfont} will choose the default font for Persian texts of your document. 
\item \Lcs{setlatintextfont} will choose the font for Latin texts of your document. If you do not use this command at all, the default \TeX\ font (fonts used in this documentation) will be used for Latin texts of your document.
\item \Lcs{setmathdigitfont} will choose the Persian font for digits in math mode. By default, digits in math mode will appear in Persian form.
\item \Lcs{setmathsfdigitfont} will choose the Persian font for digits in math mode inside \Lcs{mathsf}. By using this command, digits in math mode inside \Lcs{mathsf} will appear in Persian form and if you do not use this command at all, you will get default \TeX\ font for digits in math mode inside \Lcs{mathsf} and digits appear in their original form (Western).
\item \Lcs{setmathttdigitfont} will choose the Persian font for digits in math mode inside \Lcs{mathtt}. By using this command, digits in math mode inside \Lcs{mathtt} will appear in Persian form and if you do not use this command at all, you will get default \TeX\ font for digits in math mode inside \Lcs{mathtt} and digits appear in their original form (Western).
\end{itemize}
\subsubsection{Defining Extra Persian and Latin Fonts}
\begin{BDef}
\Lcs{defpersianfont}\Lcs{CS}\OptArgs\Largb{\Larga{font name}}\\
\Lcs{deflatinfont}\Lcs{CS}\OptArgs\Largb{\Larga{font name}}
\end{BDef}
\begin{itemize}
\item With \Lcs{defpersianfont}, you can define extra Persian fonts.
\begin{lstlisting}[numbers=none]
\defpersianfont\Nastaliq[Scale=1]{IranNastaliq}
\end{lstlisting}
In this example, we define \Lcs{Nastaliq} to stand for IranNastaliq font.
\item With \Lcs{deflatinfont}, you can define extra Latin fonts.
\begin{lstlisting}[numbers=none]
\deflatinfont\junicode[Scale=1]{Junicode}
\end{lstlisting}
In this example, we define \Lcs{junicode} to stand for Junicode font.
\end{itemize}
\subsubsection{Choosing Persian Sans Font}
\begin{BDef}
\Lcs{setpersiansansfont}\OptArgs\Largb{\Larga{font name}}\\
\Lcs{persiansffamily}\quad\Lcs{textpersiansf}\Largb{\Larga{text}}
\end{BDef}
\subsubsection{Choosing Persian Mono Font}
\begin{BDef}
\Lcs{setpersianmonofont}\OptArgs\Largb{\Larga{font name}}\\
\Lcs{persianttfamily}\quad\Lcs{textpersiantt}\Largb{\Larga{text}}
\end{BDef}
\subsubsection{Choosing Persian Iranic Font}
\begin{BDef}
\Lcs{setiranicfont}\OptArgs\Largb{\Larga{font name}}\\
\Lcs{iranicfamily}\quad\Lcs{textiranic}\Largb{\Larga{text}}
\end{BDef}
\subsubsection{Choosing Persian Navar Font}
\begin{BDef}
\Lcs{setnavarfont}\OptArgs\Largb{\Larga{font name}}\\
\Lcs{navarfamily}\quad\Lcs{textnavar}\Largb{\Larga{text}}
\end{BDef}
\subsubsection{Choosing Persian Pook Font}
\begin{BDef}
\Lcs{setpookfont}\OptArgs\Largb{\Larga{font name}}\\
\Lcs{pookfamily}\quad\Lcs{textpook}\Largb{\Larga{text}}
\end{BDef}

\subsubsection{Choosing Persian Sayeh Font}
\begin{BDef}
\Lcs{setsayehfont}\OptArgs\Largb{\Larga{font name}}\\
\Lcs{sayehfamily}\quad\Lcs{textsayeh}\Largb{\Larga{text}}
\end{BDef}

\subsubsection{Choosing Latin Sans Font}
\begin{BDef}
\Lcs{setlatinsansfont}\OptArgs\Largb{\Larga{font name}}\\
\Lcs{sffamily}\quad\Lcs{textsf}\Largb{\Larga{text}}
\end{BDef}
\subsubsection{Choosing Latin Mono Font}
\begin{BDef}
\Lcs{setlatinmonofont}\OptArgs\Largb{\Larga{font name}}\\
\Lcs{ttfamily}\quad\Lcs{texttt}\Largb{\Larga{text}}
\end{BDef}
\end{document}